RESEARCH ARTICLE



Environmental pollution, energy import, and economic growth: evidence of sustainable growth in South Africa and Nigeria

Adeolu O. Adewuyi¹ • Olabanji B. Awodumi^{2,3}

Received: 29 August 2020 / Accepted: 27 October 2020 / Published online: 19 November 2020 © Springer-Verlag GmbH Germany, part of Springer Nature 2020

Abstract

Simultaneous achievement of sustainable development goals (SDGs), especially energy efficiency (SDG 7), economic growth (SDG 8), and pollution reduction (SDG 13), has been a major challenge among the developing countries. Besides, there is absence of a study which quantifies clean production (net growth-emission effects) and energy import efficiency (net growth-energy import effects) as indicators of sustainable growth. Thus, this current study examines this issue for South Africa and Nigeria at aggregate and sectoral levels between 1981 and 2015 using simultaneous equation models and threshold regression analysis. Evidence of the sustainable growth cannot be established for South Africa and Nigeria with and without structural break. Further, the analysis shows that, with respect to Nigeria, keeping petroleum import per capita above the respective threshold enhances the environmental quality as aggregate and sectoral outputs increase. However, the CO2 emission can only induce increased GDP per capita when the petroleum import is below the threshold level. In South Africa's case, although maintaining petroleum import beyond the threshold may increase CO2 emission per capita associated with high aggregate output per capita, such emission exhibits a negligible reverse impact on output per capita. Results are found to vary across both lower and upper threshold regimes for sectors. Policy recommendations are discussed in the conclusion.

Keywords Environment \cdot Energy import \cdot Aggregate and sectoral growth \cdot Simultaneous equation \cdot Threshold analysis \cdot Nigeria and South Africa

Introduction

Countries all over the world are now struggling to achieve sustainable development (Rosen 2009; and Oyedepo 2012). This aspiration has led to the articulation of the Millennium Development Goals (MDGs) which has been transformed into sustainable development goals (SDGs). Sustainable development can be referred to as development that takes care of the requirements of the current generation without jeopardizing

Responsible editor: Nicholas Apergis

Adeolu O. Adewuyi aoadewuyi@yahoo.co.uk

Olabanji B. Awodumi benawodumi@gmail.com

- ¹ Department of Economics, University of Ibadan School of Business, Ibadan, Nigeria
- ² Nigerian Institute of Social & Economic Research (NISER), Ibadan, Nigeria
- ³ Nigerian Institute of Social and Economic Research, Ibadan, Nigeria

the ability of the upcoming generation to satisfy their own needs (United Nations 1987). The 2002 World Summit on Sustainable Development identified social, economic, and environmental sustainability as the three pillars of sustainable development (Oyedepo 2012). A combination of two of the pillars (economic and environmental sustainability) forms sustainable growth which means long-run growth achieved with energy efficiency and low carbon development (clean production). Thus, the general assumption is that growth should contribute to energy resource conservation and environmental upgrading, while energy and environment should support growth (Zuo and Ai 2011; Hu et al. 2011; Omer 2008). This means that, while energy and environment support growth, the benefits generated should reflect in energy adequacy (saving) and improvement in environmental quality (Fang et al. 2020).

There have been debates in the literature concerning whether the interaction among growth, energy, and environment is complementary or trade-off. A school of thought argued that factors substitution and technical change can successfully downgrade the role of resources (energy and environment) in economic growth (Solow 1993, 1997; Bretschger 2005). However, another school of thought (ecological economists) stated that the substitution between inputs (capital, labor, and resources) and technical advancement have limited role to play in addressing the resources scarcity problem (Stern 1997; Cleveland et al. 1984; and Hall et al. 2003). According to Hall et al. (2003), most productivity growth arose from increased energy consumption, while technical change engendered by innovations largely increase productivity via consumption of more energy. A balanced view is that market-oriented energy and environmental policies can induce the introduction of new technologies that reduce energy utilization and environmental degradation during production process (Porter and van der Linde 1995, 1999).

The seventeen sustainable development goals (17 SDGs) include low energy consumption (SDG 7), economic growth (SDG 8), and reduction in impact of climatic change (SDG 13). The need to achieve complementarity among the three SDGs was stressed in a recent special report of the Intergovernmental Panel on Climate Change on Global Warming (IPCC 2018). In the major industrialized countries in Africa, particularly South Africa and Nigeria, a significant proportion of refined petroleum products consumed are imported.1 For instance, in Nigeria and South Africa, the total import of refined petroleum products was generally higher during 2000–2015 than the earlier years.² In 2009 alone, the import of refined petroleum was greater than the domestic output of refined petroleum in Nigeria by about 225%, though this difference fell to about 78% in 2012 (US Energy Information Administration (USEIA) 2010). Also, fuel imports in South Africa and Nigeria reached about 15% and 20% of total merchandise imports (respectively) in 2012. During the same period, the gross domestic product per capita in these countries was relatively higher than the earlier years (World Bank, on-line).³ In contrast to the observed upward trends in petroleum import per capita and GDP per capita in the two countries, carbon emissions per capita fell marginally from 10.36 metric tons in 1984 to 9.15 metric tons in 2015. However, despite the fact that the average petroleum import per capita in South Africa (39.4 l) and Nigeria (43.7 l) is close during the study period, the mean GDP per capita in South Africa (US \$6625.40) is five times greater than that of Nigeria (US \$1639.80), while the mean carbon emission per capita in South Africa (9265.85 k tones) is fifteen times higher than that of Nigeria (606.13 tons). This suggests differences in the growth-energy-environment links in the two countries. There is therefore the need for empirical investigation to ascertain the links as claimed by Lin and Agyeman (2019). It should be stated that earlier related studies on Nigeria and South Africa (such as Maji 2015, Ackah 2015 and Adejumo 2019) focused on the role of either total or clean energy on productivity (economic growth) or the growth-emission link but ignored the potential bidirectional link between the variables, the particular role of the import component of the total energy, and the link of these variables with the environment.

In the light of the above, the emerging issues are as follows: (1) what is the nature of output-energy (import)-emission nexus in South Africa and Nigeria? (2) Are there sectoral differences in the output-energy (import)-emission nexus in these countries? (3) To what extent has energy import contributed to sustainable growth in the countries? Despite the importance of these policy questions, little or no study has addressed them over time.

This article highlights a number of shortcomings in the literature on the output-energy-growth nexus and contributes in the following ways. First, although there are many studies on the link between energy variety (including oil), growth, and emission (see Adewuyi and Awodumi 2017a, b for detailed review), there is dearth of specific studies on the energy import-growth-emission nexus.⁴ This represents a major deficiency in related literature. Although few studies exist on drivers of energy import, such studies show an overwhelming concentration on crude oil, but most crude oil importing countries refined the crude for re-exporting with only a fraction of the product retained for economic activities. Thus, it may be difficult to measure the net effect of the crude oil import on growth without considering the petroleum import. Only four studies are identified for Africa (Ziramba 2010 for South Africa; Marbuah 2017 for Ghana; Adewuyi 2016 and Adewuyi and Awodumi 2020 for Nigeria) despite the heavy dependent of most developing countries, especially African countries, on energy (crude or refined petroleum) import in production activities. Second, most of the existing studies analyzed the drivers of energy import in which income or economic growth plays a key role but did not recognize the fact that energy import could also propel growth. Thus, such analysis could lead to simultaneity bias which could only be solved using simultaneous equitation model. Besides, the role of energy import in pollution was neglected, while the possibility of the effect of structural changes (breaks) was also ignored. Causality analysis undertaken in most of these studies (including Adewuyi and Awodumi 2020) has been based

¹ Some oil-rich economies including Nigeria where the refinery facilities are either inadequate or underutilized, crude oil is being exported to earn foreign exchange, part of which is also used to finance the import of refined petroleum products to bridge the domestic energy gap and obtain the required energy to propel economic activities

² Due to malfunctioning of refineries and increased energy requirement in the economy (Tajudeen 2015; and Adewuyi 2016)

³ World Development indicators

⁴ See Adewuyi and Awodumi (2017a, b) for a comprehensive review of energy-output-emission linkages. It should be mentioned that studies on impact of imported inputs on productivity and output growth (Lee 1995; Busse and Groizard 2008; Kasahara and Rodrigue 2008; Goldberg et al. 2010; Halpern et al. 2015; and Olper et al. 2017) considered only capital goods and total raw material

on bivariate framework which is not rigorous given the complexity of economic relationship, while threshold analysis that can aid policy analysis was not conducted. Moreover, these studies focused on only aggregate analysis but failed to verify the sectoral links which limit policy lessons that could be drawn. Above all, a major shortcoming in the literature is the absence of a study (to the best of our knowledge) that quantifies net growth-emission effects (clean production) and net growth-energy import effects (energy import efficiency) as indicators of sustainable growth.

Fourth, empirical estimates could aid the projection of future energy import demand required to attain specific growth and pollution targets set for the economies as contained in the development policy and planning documents. They could also help in the evaluation of the energy efficiency of the economy and also provide a basis for analyzing the potential impact of public policies (such as government energy subsidy program) on aggregate and sectoral outputs and emissions. Given the association between energy consumption, output, and environmental pollution, this study would be helpful in the assessment of the level of achievement of some SDGs and development of policies for managing the negative environmental effects of energy import used in the production activities.

This present study, therefore, quantifies and analyzes the link between refined petroleum import, economic growth, and carbon emission in South Africa and Nigeria at aggregate and sectoral levels using the most recent multivariate Granger causality method and simultaneous equation model which correct for structural breaks, while threshold analysis was also done to aid policy analysis. This quantification and analysis enable us to provide robust evidence of lack of clean production and energy import efficiency in South Africa and Nigeria.

The rest of this study is organized in a way such that Section 1.2 presents stylized facts on petroleum import, economic growth, and carbon emissions in South Africa and Nigeria while Section 2 contains the review of literature. Section 3 covers the theory and methodology employed in the study and Section 4 presents and discusses the empirical results. Section 5 summarizes and concludes the article with policy implications.

Literature review

Theoretical literature

It can be stated that the existing theoretical explanation of the nexus among energy consumption, economic growth, and carbon emissions exist in pairs. The literature isolates the growthenergy (import), growth-environmental quality, and energy (import)-environmental quality links to provide partial description of the complex relationship among the variables (Adewuyi and Awodumi 2020).

On the growth-energy (import) nexus, four hypotheses dominate the literature: the growth, conservation, feedback, and neutrality hypotheses (Payne 2009; Ozturk 2010; Omri 2014). According to the growth hypothesis, unidirectional causality runs from energy consumption to economic growth, following its critical role in promoting economic growth. The conservation hypothesis suggests that a unidirectional causality run from economic growth to energy consumption, as increase in real per capital income raises the level of economic activities necessitating high energy use. Also, while the feedback hypothesis indicates bidirectional causality between energy consumption and economic growth, neutrality hypothesis implies absence of causal link between them with relatively negligible influence on each other. In addition, Bahmani-Oskooee and Economidou (2009) is among the literature that discusses the import-led growth hypothesis, which is on the role of imports in creating opportunities for growth through import-induced access to affordable inputs including energy. The growth-led import hypothesis is also about the importance of growth in facilitating imports including inputs such as energy.

The relationship between economic growth and environmental quality is popularly explained using the environmental Kuznets Curve (EKC) hypothesis. The hypothesis establishes that environmental pollution rises with income at low level of per capita income but declines as per capita income increases beyond a certain point (Grossman and Krueger 1995). Recent applications of this hypothesis include Boubellouta and Kusch-Brandt (2020) for EU28 + 2 countries, Churchill et al. (2020) for Australia, and Halliru et al. (2020) for the Economic Community of West African States. Environmental quality can also promote or hinder growth via its influence on the critical production inputs such as labor and capital. For instance, negative environmental effect (pollution) on human health (illness) will affect productivity and growth (Adewuyi and Awodumi 2017b).

With respect to energy (import)-environmental quality link, it should be stated that, while some energy products such as fossil fuel and coal generate pollution, others such as hydroelectricity, wind, and solar are environmentally friendly (clean) energy sources. Thus, environmental quality depends on the type of energy use. However, considering the harmful effect of pollution, the less the demand for energy source engenders high quantity of carbon emissions if there are equally affordable alternatives (Adewuyi and Awodumi 2017b). Given that some energy products are imported, it is imperative to state that trade facilitates access to some energy products such as crude oil and refined petroleum products. Therefore, in an energy import-dependent country, the nexus between energy and environment is a function of availability of energy via trade. The trade (import)-environment nexus is summarized under the Pollution-Haven hypothesis where countries import energy products that generate unfavorable effect on their environment, and Pollution-Halo hypothesis in which energy-related imports contribute to clean production and environment (Aller et al. 2015; Aklin 2016; Cherniwchan 2017). Going by the Heckscher-Ohlin trade theory, comparative advantage and pattern of trade lie in resource abundance. Thus, in some developing countries, the environment (natural resources) dictates pattern of trade (export of pollutionintensive primary product), while pattern of trade (import of refined or manufactured goods) also dictates the environment quality (measured by quantity of pollution). Thus, these developing countries export and import pollution-intensive goods (Copeland and Taylor 2005). The comprehensive linkage among growth, energy, and environment is still missing in the theoretical literature, which informs the partial analysis of this relationship in the empirical literature.

The theoretical analysis of the energy-growth-emission nexus can also be discussed based on the concept of sustainable economic development and the debates on the role of energy in production. As stated in the "Introduction" section of this study, there are three pillars of sustainable economic development: social, economic, and environmental sustainability (World Summit on Sustainable Development, 2002; and Oyedepo 2012). Alignment of two of the pillars (economic and environmental sustainability) generates sustainable growth which implies attainment of long-run growth with energy efficiency and low carbon development. Thus, while energy and environment foster growth, the gains obtained should lead to energy adequacy and improvement in environmental quality (Zuo and Ai 2011; Hu et al. 2011; Omer 2008; Fang et al. 2020). The debates on the significance of energy in production have been well summarized in the "Introduction" section.

Empirical literature

A number of studies have been conducted, especially within the last decade, across countries and regions on the link of energy consumption (renewable and non-renewable) with economic growth or carbon emission or both.⁵ Only few studies specifically considered the role of oil consumption (combination of locally produced and imported) in the trivariate relationship (see Adewuyi and Awodumi 2017a for studies reviewed) while those that covered oil import (crude or refined) focused on only the drivers of such import. First, specific studies on Nigeria and South Africa are reviewed. Second, a general review of the literature is conducted consisting of studies on oil-growth nexus, oil-growthemission nexus, and lastly, studies on drivers of oil import.

Literature on Nigeria and South Africa

The growth-energy (import)-emissions nexus has also been investigated in Sub-Saharan Africa, particularly Nigeria and South Africa. In Nigeria, Adewuyi and Awodumi (2017b) estimated a simultaneous equation model to establish a complete feedback effects among biomass energy, carbon emissions, and economic growth between 1980 and 2010. Subsequently, Halliru et al. (2020) employed panel quantile regression technique for six West African countries (including Nigeria) and reported a U-shaped relationship between carbon emission and economic growth over 1970-2017. In a similar analysis, Awodumi and Adewuyi (2020) adopted nonlinear autoregressive distributed lag technique (NARDL) to show that both positive and negative changes in petroleum consumption reduce economic growth but promote environmental quality during the 1980-2015 period. In contrast, autoregressive distributed lag technique (ARDL) estimates of Adejumo (2019) reveal that economic growth worsens environmental quality in the country between 1970 and 2014, while the reverse effect is negligible. Adewuyi and Awodumi (2020) found no evidence of bivariate Granger causality among energy resource import, carbon emissions, and Nigeria's economic growth during 1980-2019. Further analysis of import of petroleum products by Adewuyi (2016) indicates a positive role of economic growth between 1984 and 2013.

In the case of South Africa, Toda and Yamamoto Granger causality test results employed by Ziramba (2009) suggest feedback causal links between oil consumption and industrial production. Moreover, Ziramba (2010) utilized data ranging from 1980 to 2006 in a PMG panel ARDL analysis to show that import demand for crude oil is positively affected by economic growth in the country. Using Pairwise Dumitrescu-Hurlin Panel causality approach for the same country, Azam (2019) found significant bidirectional causality between growth and each of energy consumption and carbon emissions, but a unidirectional causality running from energy consumption to carbon emissions. Granger causality estimates of Bildirici and Bakirtas (2014) found neutral link between oil (and natural gas) consumption and economic growth in the country between 1980 and 2011.

In Sub-Saharan Africa, including Nigeria and South Africa, Bah et al. (2020) confirmed the inverted U-shaped link between growth and environmental pollution (EKC hypothesis) during 1971–2012. FMOLS estimates of Behmiri and Manso (2013) reveal significant feedback effect between carbon emission and economic growth for South Africa, but neutral in the case of Nigeria. Further, Mensah et al. (2019) PMG panel ARDL estimation method provides evidence of bidirectional causal link between fossil fuel energy consumption and each of economic growth and carbon emissions among SSA countries.

⁵ For excellent and comprehensive review of these studies, see Adewuyi and Awodumi (2017a) and Yao et al. (2019)

It is clear from the above review that, although some studies exist on South Africa and Nigeria, only three of such studies cover energy import (Ziramba 2010 for South Africa; Adewuyi 2016 & Adewuyi and Awodumi 2020 for Nigeria). Besides, Adewuvi and Awodumi (2020), which extended analysis of output-emission linkage to energy import, is based on bivariate Granger causality framework that cannot reveal direct, indirect and joint effects of the variables on each other. This study is also limited in terms of coverage or scope (only aggregate without sectoral analysis) and robustness of analysis which can be achieved using complementary approaches such as multivariate Granger causality, simultaneous equation model, and threshold regression analysis as evident in this study. It is also limited because of its inability to quantify indicators of sustainable growth (clean production and energy import efficiency) which is part of the focus of this present study.

General studies on oil (import)-growth-emission nexus

Empirical literature provides considerable support for the critical role of oil consumption in the growth process. For instance, ECM-Granger causality approach of Zou and Chau (2006) revealed unidirectional causality running from oil consumption to economic growth (growth hypothesis) in China between 1953 and 2002. Using the impulse response function and bootstrap corrected causality methods, Pereira and Pereira (2010) found that oil, as well as coal and natural gas, caused economic growth in Portugal during the 1977-2003 period. These findings are further confirmed among a group of 42 countries by Halkos and Tzeremes (2011), in Cameroon by Wandji (2013), among the G-6 countries by Chu and Chang (2012), a panel of 49 countries by Chu (2012) and Akhmat and Zaman (2013) for the case of South Asia. Similar reports were made in the short-run by Behmiri and Manso (2014) (Latin America) and Lach (2015) (Poland). It should be stated that few studies also showed that causality runs from economic growth to oil consumption rather than the other way round. These include Ali and Harvie (2013), Lean and Smyth (2014), and Zheng and Luo (2013) for Libya, Malaysia, and China, respectively.

Irrespective of the time horizon, some studies confirmed the feedback hypothesis between oil consumption and economic growth. This hypothesis is evident in studies conducted for China (Yuan et al. 2008; and Zhang and Yang 2013), the 27 OECD countries (Behmiri and Manso 2012a), Portugal (Behmiri and Manso 2012b), Eurasia (Bildirici and Kayıkçı 2013), BRICS countries (Azam 2019), Saudi Arabia (Alkhathlan and Javid 2013), Malaysia (Saboori and Sulaiman 2013; and Park and Yoo 2014), the Philippines (Lim et al. 2014), and G20 (Pao and Chen 2018).

Few studies also investigated the carbon emission implication of the link between growth and oil consumption. AlMulali (2011) and Apergis and Pavne (2014) showed evidence of a bidirectional causality between carbon emission and economic growth among MENA countries and 7 Central American countries, respectively. In Wuhan, China, Rao and Yan (2020) demonstrated that the interaction between industrial waste gas emissions and industrial wastewater emissions is detrimental for economic growth. This finding is not confirmed among the European Union countries where Balsalobre-Lorente and Leitão (2020) discovered positive influence of carbon emission on growth, following climate change and greenhouse gas implication of high industrial activities. However, Bloch et al. (2012) reported that causality only runs from growth to carbon emission in China, a finding validated in the short-run among 10 MENA countries by Farhani and Shahbaz (2014). Thus, Ali et al. (2020) reported a U-shaped link between economic growth and environmental degradation among European countries, but Dong et al. (2020) confirmed reducing effect of growth on carbon emission in China's industrial sector.

For OECD and the selected 22 African countries, Shafiei and Salim (2014) and Mensah et al. (2019) revealed that the link between non-renewable energy consumption and carbon emission (as well as growth) is bidirectional. A number of studies however proved that non-renewable energy caused carbon emission with no feedback effect. This result was observed for EU countries by Dogan and Sekan (2016a) and Bolük and Mert (2014), top renewable energy countries by Dogan (2015).

Most of the studies on oil import have shown overwhelming interest on the drivers of oil import (including growth) but ignored its relationship with carbon emission. However, there appears to be a consensus in the energy import literature on the significant role of income or economic growth in oil import demand, which is evident across the few available studies. Although studies have largely concentrated on Asian economies, some works have been done in few other regions. For example, Zhao and Wu (2007) revealed that industrial and transport output growth has significant long-run positive impact on crude oil import in China during the year 1995-2006. Also, Ghosh (2009) showed that income has positive and unidirectional causal effect on crude oil import in India between 1970 and 2006. Kim and Baek (2013) obtained similar results in the case of Korean during the 1986-2010 period, although their results indicated that oil price is more important in the short-run. Also, the simulation results of Adams et al. (2000) portrayed that the economic growth has positive effect on crude oil import in Thailand during 1992-2010. However, Mardiana et al. (2013) reported that for the case of Indonesia, oil import (crude oil and petroleum product) is more affected by transportation fuel consumption while the role of economic growth is insignificant.

Camacho-Gutiérrez (2010) provided an empirical evidence that the US import demand for Mexican crude oil responds positively to changes in income between 1990 and 2010. Moore (2011) found support for this result in Barbados from 1998 to 2009. This positive effect of economic growth on crude oil import has also been validated in West Africa. For instance, Marbuah (2017) found that income has increasing impact on crude oil import in Ghana from 1980 to 2012. Also, Altinay (2007) showed that income has both short-run and long-run impacts on import demand for crude oil in Turkey during the 1980–2005 period, while for the same country, Ozturk and Arisoy (2016) obtained related results after analyzing data covering 1966–2012. In contrast to this finding, Ediger and Berk (2011) found that changes in crude oil imports in the same country are not significantly affected by economic growth during 1968-2007. The findings for Spain appear to contrast markedly from those obtained for other countries. This is evident in the analysis conducted by Balaguer et al. (2015), where unidirectional causality runs from aggregate energy import to economic growth from 1900 to 2012. Further, Fedoseeva and Zeidan (2018) reported income as the most significant driver of European energy imports.

In conclusion, the foregoing shows that the existing studies in this area suffer from a number of shortcomings. The studies on bivariate oil-growth nexus and the few studies on trivariate oil-growth-emission link are inconclusive given their diverse findings and concentrations in Asia and developed countries, with very few studies on Africa. Also, literature on energy import is just developing as shown by the review while only four studies are identified for Africa (Ziramba 2010 for South Africa; Marbuah 2017 for Ghana; Adewuyi 2016 & Adewuyi and Awodumi 2020 for Nigeria) despite the heavy dependent of most developing countries, especially African countries, on energy (crude or refined petroleum) import in production activities. Most studies analyzed the drivers of energy import in which income or economic growth plays a key role but did not recognize the fact that energy import could also propel growth. Thus, such analysis could lead to simultaneity bias and unreliable model estimates. Besides, the role of energy import in pollution was neglected, while the possibility of the effect of structural breaks was also ignored. Causality analysis conducted in most of these studies has been based on bivariate framework which is not robust given the complexity of economic relationship, while threshold analysis that can aid policy analysis was not conducted. Moreover, these studies focused on only aggregate analysis but failed to investigate the sectoral links which limit policy implications that could be drawn. There is no study that quantifies net growth-emission effects (clean production) and net growth-energy import effects (energy import efficiency) as indicators of sustainable growth.

This present study therefore fills (or addresses) the above highlighted gaps (or shortcomings) by quantifying and analyzing the link among refined petroleum import, economic growth, and carbon emission in Nigeria and South Africa at aggregate and sectoral levels using the most recent multivariate Grange causality approach and simultaneous equation model (correct for structural breaks) with threshold regression analysis. The above discussions are summarized in Table 10 in Appendix 1.

Theory and methods

Theory

According to sustainable development literature, a sustainable economic growth can be achieved only when the relative contribution of income or output to achieving energy efficiency and improved environmental quality (clean production) is more than the relative contributions of input of energy and environment to output (Zuo and Ai 2011; Hu et al. 2011; Omer 2008; Yu et al. 2006; Yang et al. 2004). This implies that the net effect of growth-emission nexus leads to fall in emission (low carbon development-clean production) combined with the net effect of growth-energy nexus which should be a reduction in energy use (energy conservation-energy efficiency). Thus, there is a need for an empirical investigation of the energy-growth-pollution nexus through which the validity of the foregoing can be ascertained for the countries of interest in this study. Investigating the energy (import)-growth-pollution nexus requires an integrated approach that combines endogenous growth model with energy import demand and pollution production models.

Growth model

The endogenous growth model emphasizes the role of human capital development [L(H)—knowledge] apart from physical labor and capital (L and K—primary inputs) as well as technological efficiency (A) in driving long-run real economic (income) growth (Y) (Romer 1996; and Jones and Manuelli 1997).

$$Y = A^*K^*L^*L(H) \tag{1}$$

Equation 1 can be expressed in per capita terms as follows:

$$y = a^*k^*H \tag{2}$$

where y = output per capita; k = investment per capita or capital-labor ratio; H = human capital development; and a =productivity growth or efficiency parameter, which is driven by some factors including imported petroleum per capita (em) and environmental quality (carbon emission per capita—co₂), financial development (*F*), and trade openness (*T*). Productivity growth or efficiency parameter can be given as follow:

$$\mathbf{a} = \mathbf{f} \ (\mathbf{em}, \mathbf{co}_2, \mathbf{F}, \mathbf{T}) \tag{3}$$

Imported refined petroleum (em) has been seen as either a critical intermediate input in the production process (Shahbaz et al. 2013; Dasgupta and Roy 2015) or a facilitator of efficiency of primary inputs (Adewuyi and Awodumi 2017b). Environmental quality has also been argued to facilitate productivity and efficiency (or otherwise) of the primary inputs depending on the level of carbon emission (co₂) from production and its effects on human (labor) health. Other drivers of productivity and long-run growth are human capital development (measured as % of primary school enrolment in gross enrolment-H), financial development (measured as % of private sector credit in total credit -"F"), and trade openness (measured as % of total trade in GDP -"T") (Shahbaz et al. 2013; Dasgupta and Roy 2015; and Adewuyi and Awodumi 2017b).

Substituting for "a" in Eq. 2, we have Eq. 4; we have an extended growth model expressed in per capita form as follows:

$$\mathbf{y} = \mathbf{em}^* \mathbf{co}_2 \mathbf{k}^* \mathbf{F}^* \mathbf{T}^* \mathbf{H} \tag{4}$$

In the energy-growth literature, four hypotheses are commonly used to describe the link between economic growth and energy consumption. These are the conservation hypothesis, the growth hypothesis, the feedback hypothesis, and the neutrality hypothesis (Adewuyi and Awodumi 2017a, b). The conservation hypothesis suggests that economic growth raises the level of energy consumption through increased economic activities across all sectors. In most countries, petroleum products remain the largest form of energy used in industries and other sectors, and for developing economies, these products are largely imported. Higher incomes could however create incentives for an economy to adopt alternative energy sources such as renewable energy which tends to reduce petroleum import. The growth hypothesis holds that energy consumption drives economic growth as it represents a major input in the production processes by facilitating the functioning of both capital and labor. According to the feedback hypothesis, there exists bidirectional link between growth and energy consumption while the neutrality hypothesis states that no causal link exists between energy consumption and growth.

Energy (import) demand model

Based on the imperfect substitution model with budget constraint and utility function of a representative consumer which contains two goods—local and foreign products—the constrained optimization problem will yield demand function for both local and imported goods (Goldstein and Khan 1985; Narayan and Smyth 2005; and Adewuyi 2016). Summing import demand function of all consumers in the economy gives an aggregate import demand function in which the quantity of import demand (M) is a direct function of the level of aggregate real income of the importing country(Y) and relative price (RP) which is the ratio of import price and domestic price of substitutes or import competing goods.

$$\mathbf{M} = \mathbf{Y}^* \mathbf{R} \mathbf{P} \tag{5}$$

Assuming that import demand includes energy products, therefore, energy import (EM) model in line with Adewuyi (2016) and Mensah et al. (2016) is specified as follows:

$$\mathbf{E}\mathbf{M} = \mathbf{Y}^* \mathbf{R} \mathbf{P} \tag{6}$$

This means that the aggregate energy import (EM) demand depends on real income level (Y) and relative price (RP). Thus, high real income encourages energy (petroleum) import while high relative price discourages it. Energy (import) demand model accounts for the role of physical capital (K) which requires energy to function; financial capital (F) which is required to finance import; and environmental conditions (CO2 emissions) which determines the kind of energy used (Mensah et al. 2016; and Adewuyi and Awodumi 2017b). We have the following equation in per capita term:

$$em_{t} = y_{t} * RP_{t} * k_{t} * F_{t} * co_{2t}$$
(7)

Pollution (carbon emissions) production model

The consideration of carbon emission implication of the energy-growth nexus becomes imperative as economic growth largely rests on increased energy (petroleum) consumption. Output (goods) and carbon emission (bads) are joint output in the production process (Adewuyi and Awodumi 2017b). Also, a high carbon emission has been variously associated with a high fossil fuel consumption required to compliment primary inputs (Apergis et al. 2018; Ma et al. 2019). Thus, the production of carbon emission (CO2) depends on the level of real economic activities (output) or income (Y); petroleum consumption, which is mainly imported (EM); physical capital (K); and financial development (F) and trade openness (T).

$$CO2_t = Y_t * EM_t * K_t * F_t * F$$
(8a)

In per capita terms, Eq. 8a can be written as:

$$\mathbf{co}_{2t} = \mathbf{y}_t * \mathbf{em}_t * \mathbf{k}_t * \mathbf{F}_t * \mathbf{T}_t \tag{8b}$$

Energy (import)-growth-emission nexus

A simultaneous equation model can then be specified for the link between energy import, economic growth, and environmental quality (pollution) as follows:

$$\mathbf{y} = \mathbf{k}^* \mathbf{e} \mathbf{m}^* \mathbf{H}^* \mathbf{F}^* \mathbf{T}^* \mathbf{c} \mathbf{o}_2 \tag{9}$$

 $em_{t} = y_{t} * RP_{P} * k_{t} * F_{t} * co_{2t}$ (10)

 $co_{2t} = y_t * em_t * k_t * F_t * T_t$ (11)

Methods

Following from the foregoing theoretical framework, the study first adopts multivariate linear and nonlinear Granger causality to establish the causal relationship among the variables. It thereafter employs simultaneous equation approach as well as dynamic threshold technique to provide more robust estimates of the empirical analysis.

Multivariate linear and nonlinear Granger causality test

This study employs the most recent multivariate Granger causality method by Bai et al. 2018 which is a refinement of the previous approach discussed in Bai et al. (2011). It was noticed that the central limit theory prescribed by Hiemstra and Jones (1994) and enlarged by Bai et al. (2010) as well as applied by Bai et al. (2011) is invalid for statistical inference (Bai et al. (2018)). This led to its re-estimation of the probabilities and reestablishment of the central limit theory that support the new test statistic developed by Bai et al. (2018).

Starting with Bai et al. (2011), the multivariate linear Granger causality method can be used to test the linear relationship between two vectors of different stationary time series say $X_t = (x_1, t, ..., x_{n1,j})$ and $Y_t = (y_1, t, ..., y_{n2,j})$, where there are $n_1 + n_2 = n$ series in total, and then the following *n*-equation VAR system can be applied as follows:

VAR framework

$$\begin{pmatrix} x_t \\ y_t \end{pmatrix} = \begin{pmatrix} A_{x[n_1 \times 1]} \\ A_{y[n_2 \times 1]} \end{pmatrix} + \begin{pmatrix} A_{xx}(L)[n_1 \times n_1] & A_{xy}(L)[n_1 \times n_2] \\ A_{yx}(L)[n_2 \times n_1] & A_{yy}(L)[n_2 \times n_2] \end{pmatrix} (12)$$
$$+ \begin{pmatrix} x_{t-1} \\ y_{t-1} \end{pmatrix} + \begin{pmatrix} e_{x,t} \\ e_{y,t} \end{pmatrix}$$

where $A_{x[n_1 \times 1]}$ and $A_{y[n_2 \times 1]}$ are two vectors of intercept terms; $A_{xx}(L)[n_1 \times n_1]$, $A_{xy}(L)[n_1 \times n_2]$, $A_{yx}(L)[n_2 \times n_1]$, and $A_{yy}(L)[n_2 \times n_2]$ are matrices of lag polynomials and $e_{x,t}$ and $e_{yz,t}$ are corresponding residual terms.

In order to test the multivariate causal relationship between vectors X and Y, the following null hypotheses H_0^{-1} : $A_{xy}(L) = 0$, H_0^{-2} : $A_{yx}(L) = 0$ and both are tested and there are four different situations for the causality relationships which include:

- 1) Unidirectional causality occurs from Y to X if H_0^{-1} is rejected but H_0^{-2} is not rejected.
- 2) There is unidirectional causality from X to Y if H_0^2 is rejected but H_0^{-1} is not rejected.
- 3) Feedback relation occurs when both H_0^{1} and H_0^{2} are rejected.
- 4) X_t and Y_t are not related (independent) when both H_0^{-1} and H_0^{-2} are not rejected.

Meanwhile, if the time series are cointegrated, then a restricted VAR (ECM-VAR) is employed to test the causality relationship between two vectors of non-stationary time series. In doing so, let $X_t = (X_1, t, ..., X_n, t)$ and $Y_t = (y_1, t, ..., y_n, t)$ and let $X_{it} = X_{it}$ and $Y_{it} = Y_{it}$ be the corresponding stationary differencing series such that there are $n_1 + n_2 = n$ series in total. Thereafter, if X_t and Y_t are cointegrated, then the ECM-VAR model below can be adopted:

ECM-VAR framework

$$\begin{pmatrix} x_t \\ y_t \end{pmatrix} = \begin{pmatrix} A_{x[n_1 \times 1]} \\ A_{y[n_2 \times 1]} \end{pmatrix} + \begin{pmatrix} A_{xx}(L)[n_1 \times n_1] & A_{xy}(L)[n_1 \times n_2] \\ A_{yx}(L)[n_2 \times n_1] & A_{yy}(L)[n_2 \times n_2] \end{pmatrix} (13)$$
$$+ \begin{pmatrix} x_{t-1} \\ y_{t-1} \end{pmatrix} + \begin{pmatrix} \alpha_x[n_1 \times 1] \\ \alpha_y[n_2 \times 1] \end{pmatrix} ecm_{t-1} + \begin{pmatrix} e_{x,t} \\ e_{y,t} \end{pmatrix}$$

where ecm_{t-1} is the lag 1 case error correction model parameter and $\alpha_x[n_1 \times 1]$ and $\alpha_y[n_2 \times 1]$ are vector of coefficients of ecm_{t-1} term. Equation 13 shows that causation can occur through either the lagged dynamic terms or the error correction term. Thus, one could test the null hypotheses of $H_0^{1:} A_{xy}(L) = 0$ and/or $H_0^{2:} A_{yx}(L) = 0$ to identify the causality relationship.

According to Bai et al. (2010) and Bai et al. (2011), multivariate nonlinear causality is also conceptually similar to bivariate nonlinear causality test developed by Baek and Brock (1992) and subsequently refined by Hiemstra and Jones (1994). In order to identify any nonlinear Granger causality relationship between two vectors of the time series, say $X_t = (X_{I_t}, t, ..., X_{n_t}t)'$ and $Y_t = (Y_{I_t}, t, ..., Y_{n_t}t)'$ in the multivariate scheme, one has to first obtain the residuals from either the VAR or ECM-VAR model above and thereafter apply a nonlinear Granger causality test to the residual series.

In the new procedure for estimating the linear (and nonlinear) Granger causality by Bai et al. (2018), all series are standardized such that a common standard deviation (1) is shared, $(Y_t - mean(Y_t))/SD(Y1t)$, where SD is the standard deviation.

Simultaneous equation model

In consistency with the theoretical framework earlier presented and the empirical literature on growth (Barro, 2003 and Shahbaz et al. 2013), energy import demand (Adewuyi 2016; Mensah et al. 2016), and environmental Kuznets curve (Grossman and Krueger 1995), this study re-specifies Eqs. 9 to 11 in log-linear and per capita terms and also captures the effect of structural break in petroleum import:

$$ln(\text{GDPPC})_{t} = \alpha_{0} + \alpha_{1}ln(CO2PC)_{t} + \alpha_{2}ln(PIMPC)_{t} + \alpha_{3}ln(PCAP)_{t} + \alpha_{4}ln(HCAP)_{t} + \alpha_{5}ln(FDEV)_{t} + \alpha_{6}ln(PIMPDU)_{t} + \varepsilon_{t}$$

$$ln(CO2DC) = \alpha_{1} + \alpha_{1}ln(CD2PC) + \alpha_{1}ln(PIMPDC) + \alpha_{1}ln(PIMPC) + \alpha_{2}ln(PIMPC) + \alpha_{2}ln(PIMPC) + \alpha_{3}ln(PIMPC) + \alpha_{4}ln(PIMPC) + \alpha_{5}ln(PIMPC) + \alpha_{5}ln(PI$$

$$ln(CO2PC)_{t} = \theta_{0} + \theta_{1}ln(GDPPC)_{t} + \theta_{2}ln(PIMPC)_{t} + \theta_{3}ln(PCAP)_{t} + \theta_{4}ln(TO)_{t} + \theta_{5}ln(FDEV)_{t} + \theta_{6}PIMPDU_{t} + \pi_{t} \quad (15)$$

$$ln(PIMPC)_{t} = \delta_{0} + \delta_{1}ln(GDPPC)_{t} + \delta_{2}ln(CO2PC)_{t} + \delta_{3}ln(EXR)_{t} + \delta_{4}ln(FDEV)_{t} + \delta_{5}ln(PCAP)_{t} + \delta_{6}PIMPDU_{t} + \mu_{t}$$
(16)

where GDPPC = GDP per capita; PIMPC = petroleum import pen capita; CO2PC = carbon emission per capita; PCAP = physical capital; HCAP = human capital; FDEV = financial development; TO = trade openness; EXR = exchange rate; and PIMPDU = petroleum import structural break dummy. ε_t , π_t , and μ_t are the stochastic terms. These variables are well defined and measured in Table 1.

The same simultaneous equation model is specified for each sector, but the output variable (GDPPC) is replaced by the respective sector's GDP such that AGDPPC, MGDPPC, and SGDPPC are the per capital output in the model for agricultural, manufacturing, and service sectors, respectively. The study adopts both three-stage least square (3SLS) and system generalized method of moments (sys-GMM) to estimate the specified simultaneous equation models. These techniques provide a robust estimator that accounts for the inherent contemporaneous correlation of the error terms and correlation among the explanatory variables (endogeneity). Owing to data availability constraint, empirical analysis in this study covers the period of 1981–2015. Also, all missing values⁶ are interpolated using 5-year average while all the series are used in their natural log form.

Threshold approach

This study further investigates the contingency effects of petroleum import on the link between carbon emissions and economic growth. The threshold regression approach suggested by Hansen (1999, 2000) is therefore adopted to explore the nonlinear behavior of carbon emission and economic growth in the presence of petroleum import. Based on the two-regime threshold regression, the following model is specified for both aggregate and sectoral analyses using petroleum import as a threshold variable:

$$GDPPC_{t} = \begin{cases} \alpha_{1} + \beta_{1}CO2PC_{t} + \mu_{t}, & I(PIMPC_{t} < \gamma) \\ \alpha_{2} + \beta_{2}CO2PC_{t} + \mu_{t}, & I(PIMPC_{t} \ge \gamma) \end{cases}$$
(17)

$$CO2PC_{t} = \begin{cases} \delta_{1} + \theta_{1}GDPPC_{t} + \pi_{t}, \ I(PIMPC_{t} < \phi) \\ \delta_{2} + \theta_{2}GDPPC_{t} + \pi_{t}, \ I(PIMPC_{t} \ge \phi) \end{cases}$$
(18)

where I(.) is 1 when the argument in parenthesis is valid, and 0 if otherwise. The threshold variable is the petroleum import per capita (*PIMPC*) which is used to categorize the sample into two regimes such that regime 1 represents the samples below the threshold and regimes 2 are those above the threshold. Also, the effect of carbon emission per capita on GDP per capita is β_1 and β_2 within the low and high regime, respectively (Eq. 17) while the impact of GDP per capita on the carbon emission per capita is θ_1 and θ_2 respectively (Eq. 18). The unknown threshold parameter is γ and ϕ in the respectively.

Similarly, by employing GDPPC as threshold variables, we have:

$$PIMPC_{t} = \begin{cases} \alpha_{1} + \beta_{1}CO2PC_{t} + \mu_{t}, & I(GDPPC_{t} < \gamma) \\ \alpha_{2} + \beta_{2}CO2PC_{t} + \mu_{t}, & I(GDPPC_{t} \ge \gamma) \end{cases}$$
(19)

$$CO2PC_{t} = \begin{cases} \delta_{1} + \theta_{1}PIMPC_{t} + \pi_{t}, \ I(GDPPC_{t} < \phi) \\ \delta_{2} + \theta_{2}PIMPC_{t} + \pi_{t}, \ I(GDPPC_{t} \ge \phi) \end{cases}$$
(20)

In the same vein, utilizing CO2PC as threshold variables leads to the following:

$$PIMPC_{t} = \begin{cases} \alpha_{1} + \beta_{1}GDPPC_{t} + \mu_{t}, & I(CO2PC_{t} < \gamma) \\ \alpha_{2} + \beta_{2}GDPPC_{t} + \mu_{t}, & I(CO2PC_{tt} \ge \gamma) \end{cases}$$

$$(21)$$

$$GDPPC_{t} = \begin{cases} \delta_{1} + \theta_{1}PIMPC_{t} + \pi_{t}, & I(CO2PC_{t} < \phi) \\ s_{1} + \theta_{2}PIMPC_{t} + \sigma_{t}, & I(CO2PC_{t} < \phi) \end{cases}$$

$$SDPPC_t = \begin{cases} \sigma_1 + \sigma_1 nn \sigma_t + n_t, n(\sigma \sigma_1 - \sigma_t - \varphi) \\ \delta_2 + \sigma_2 PIMPC_t + n_t, n(\sigma \sigma_1 - \sigma_t - \varphi) \\ (22) \end{cases}$$

The procedures for estimating the threshold regression of Hansen (1999, 2000) follow three stages. The first step involves the computation of the sum of square errors (SSE) for a given threshold. Next is the estimation of $\hat{\gamma}(and \hat{\phi})$ by minimizing

⁶ Total petroleum import is interpolated (1985, 1996, 1997, 2013 and 2014); capital input (2014)

 Table 1
 Variable description and data sources

Variable	Description	Measurement	Data sources
GDPPC	Aggregate output	GDP per capita (constant 2005 US\$)	World Bank, World Development Indicators, 2017
AGDPPC	Agricultural output	Agriculture, value added per capita (constant 2005 US\$)	World Bank, World Development Indicators, 2017
MGDPPC	Manufacturing output	Manufacturing, value added per capita (constant 2005 US\$)	World Bank, World Development Indicators, 2017
SGDPPC	Service output	Services, etc., value added per capita (constant 2005 US\$)	World Bank, World Development Indicators, 2017
PIMPC	Import of refined petroleum products	Total imports of refined petroleum products (liters per capita)	United States Energy Information Administration
PCAP	Capital input	Gross fixed capital formation per capita (constant 2005 US\$)	World Bank, World Development Indicators, 2017
HCAP	Human capital development	Primary school enrolment (% of gross enrolment)	World Bank, World Development Indicators, 2017
EXR	Exchange rate	Real effective exchange rate index $(2010 = 100)$	World Bank, World Development Indicators, 2017
CO2PC	Carbon emission	1000 metric tons per capita	World Bank, World Development Indicators, 2017
FDEV	Financial capital	Domestic credit to private sector (% of GDP)	World Bank, World Development Indicators, 2017
ТО	Trade openness	Trade (% of GDP)	World Bank, World Development Indicators, 2017
PIMPDU	Petroleum import dummy	D = 1 after break; 0 otherwise	Computed

Source: Author's compilation

the sum of squares. Consequently, an *F* test is conducted to determine the existence of a threshold effect and to test the null hypothesis of linearity (such as H_0 : $\beta_1 = \beta_2$; and H_0 : $\theta_1 = \theta_2$ in the threshold models in Eqs. 17 and 18). The *F* test is computed as:

$$F(\gamma) = \frac{SSE_0 - SSE_1(\widehat{\gamma})/1}{SSE_1(\widehat{\gamma})/n(T-1)} = \frac{SSE_0 - SSE_1(\widehat{\gamma})}{\widehat{\sigma}^2}$$
(23)

A threshold effect is said to exist if the null hypothesis is rejected. Using Hansen's (1996) heteroscedasticity-consistent Lagrange multiplier (LM) test for thresholds, Hansen (1999, 2000) proposed a "bootstrap" method to compute the asymptotic distribution statistics using likelihood ratio to test the significance of threshold effect. Specifically, since the bootstrap procedure attains the first-order asymptotic distribution, the associated *p*-values are asymptotically valid and reliable.

Empirical results and discussion

Preliminary analysis

The descriptive statistics of the variables used in the regression analysis is presented in Table 2. The statistics show that the average GDP per capita, manufacturing output per capita, and service output per capita are higher in South Africa than in Nigeria. Also, the average petroleum import per capita is higher in Nigeria (44 l) than in South Africa (39 l), but the mean level of carbon emission per capita is 9266 thousand metric tons in South Africa compared to 606 thousand metric tons in Nigeria. In terms of the variability, the GDP per capita appears to be the most volatile among the variables of interest in both countries with the standard deviation of 462.01 and 626.20 for Nigeria and South Africa respectively. In the same vein, all sectoral outputs are more volatile in South Africa than Nigeria, except the service output. Similarly, variability is observed to be larger in South Africa for petroleum import capita and carbon emission per capita than in Nigeria. The trends of aggregate and sectoral outputs, energy import, and carbon emissions for South Africa and Nigeria are presented in Figs. 7, 8, 9, and 10 in Appendix 2.

As reported in Table 3, the results of correlation analysis show weak association among explanatory variables but modest association between petroleum import per capita and agricultural output per capita for Nigeria. In the case of South Africa, a weak correlation is generally observed in most of the explanatory variables. Moreover, existence of structural break in petroleum import per capita is noticed in Nigeria and South Africa as test results indicate 2000 and 1998 as break dates, respectively. Consequently, in addition to conventional unit root tests (ADF, PP, and KPSS), unit root with structural break test (Perron 2006) and nonlinear unit root tests (KSS and Kruse) were conducted. The overall results from these tests suggest mixed level of stationarity among the series, which further informs the use of both linear and nonlinear estimation techniques, as well as SEM modelling as the series are used in the estimation accordingly (Table 4). All original unit root results are contained in Tables 11, 12, 13, 14, 15 in Appendix 3.⁷

⁷ Appendix 3: Table 11, Conventional unit root test results; Table 12, Conclusion for conventional unit root tests; Table 13, nonlinear unit root tests; Table 14, Conclusion for nonlinear unit root tests; Table 15, Unit root tests with structural breaks (Perron 2006)

Table 2 Desc	criptive statistics											
	AGDPPC	CO2PC	EXR	FDEV	GDPPC	HCAP	MGDPPC	PCAP	PIMPC	PIMPDU	SGDPPC	TO
Nigeria												
Mean	343.93	606.13	154.34	15.08	1639.80	93.03	123.63	242.32	43.70	0.51	654.53	51.13
Median	254.80	636.26	99.13	13.43	1405.89	89.63	110.98	184.88	42.70	1.00	469.02	53.03
Maximum	582.53	870.97	546.05	38.39	2548.43	112.81	250.59	775.09	74.01	1.00	1340.35	81.81
Minimum	187.11	306.82	49.74	8.71	1147.07	78.46	82.40	113.18	6.76	0.00	378.16	21.45
Std. dev.	141.74	165.98	126.71	6.17	462.01	8.71	43.52	144.51	21.05	0.51	317.65	16.59
Skewness	0.57	-0.17	1.71	2.43	0.81	0.80	1.66	1.88	-0.25	-0.06	1.08	-0.23
Kurtosis	1.59	2.00	5.01	9.08	2.11	2.92	4.99	6.85	1.82	1.00	2.56	2.07
J-Bera	4.78***	1.64	22.94*	88.38*	5.02***	3.76	21.87*	42.09*	2.39	5.83**	7.07**	1.55
Sum	12,037.50	21,214.59	5401.90	527.73	57,392.90	3256.00	4327.03	8481.03	1529.64	18.00	22,908.45	1789.40
S.S. dev.	683,047.10	936,623.70	545,861.80	1294.58	7,257,382.00	2582.09	64,383.41	710,031.60	15,071.09	8.74	3,430,575	9353.01
Obs	35	35	35	35	35	35	35	35	35	35	35	35
South Africa												
Mean	173.68	9265.85	100.95	117.45	6625.40	102.53	937.59	18.54	39.37	0.47	3884.03	Ι
Median	174.46	9213.42	99.44	116.36	6436.64	106.51	948.85	18.51	28.34	0.00	3634.27	Ι
Maximum	213.28	10,357.15	148.75	160.12	7623.13	117.34	1065.84	23.98	102.04	1.00	4760.80	Ι
Minimum	141.48	7847.14	69.45	69.93	5836.59	78.09	835.27	15.15	4.55	0.00	3341.35	Ι
Std. dev.	16.05	640.52	16.85	29.24	626.20	11.28	66.93	2.41	26.63	0.51	494.97	I
Skewness	0.11	0.07	0.43	-0.35	0.45	-1.25	0.09	0.52	0.72	0.13	0.72	Ι
Kurtosis	2.93	2.42	3.36	1.79	1.68	3.51	1.89	2.48	2.29	1.02	1.88	Ι
J-Bera	0.07	0.48	1.18	2.59	3.39	8.73*	1.67	1.81	3.42	5.33***	4.43	Ι
Sum	5557.79	296,507.20	3230.43	3758.54	212,012.90	3281.03	30,002.97	593.22	1259.99	15.00	124,288.90	Ι
S. S. dev.	7981.13	12,718,057	8804.91	26,501.4	12,156,069.0	3942.47	138,877.40	180.71	21,985.94	7.97	7,594,876.0	Ι
Obs	32	32	32	32	32	32	32	32	32	32	32	I
Source: Autho	r's computation;	*, **, and*** de	enote significanc	e at 1%, 5%, ^s	and 10% levels, re-	spectively; J-I	Bera = Jarque-Be	ra; S. S. dev = su	um square devia	ttion		

Correlation statistics

Table 2

	Telution Statis	dies -										
	AGDPPC	CO2PC	EXR	FDEV	GDPPC	НСАР	MGDPPC	PCAP	PIMPC	PIMPDU	SGDPPC	ТО
Nigeria												
AGDPPC	1.00											
CO2PC	-0.03	1.00										
EXR	-0.41	0.39	1.00									
FDEV	0.31	0.24	0.06	1.00								
GDPPC	0.91	0.08	-0.14	0.33	1.00							
HCAP	-0.32	0.67	0.60	0.03	-0.17	1.00						
MGDPPC	0.73	0.01	-0.14	0.10	0.86	-0.28	1.00					
PCAP	0.24	0.35	0.34	0.26	0.52	0.25	0.58	1.00				
PIMPC	0.70	0.49	-0.12	0.36	0.65	0.16	0.39	0.27	1.00			
PIMPDU	0.80	-0.02	-0.42	0.27	0.63	-0.13	0.36	0.02	0.66	1.00		
SGDPPC	0.95	-0.09	-0.31	0.28	0.97	-0.36	0.87	0.39	0.58	0.68	1.00	
TO	-0.03	-0.41	-0.50	0.01	-0.29	-0.26	-0.44	-0.38	-0.12	0.25	-0.19	1.00
South Africa												
AGDPPC	1.00											
CO2PC	0.41	1.00										
EXR	-0.01	0.32	1.00									
FDEV	-0.34	-0.35	-0.73	1.00								
GDPPC	0.23	0.32	-0.47	0.57	1.00							
HCAP	-0.24	-0.56	-0.39	0.56	-0.10	1.00						
MGDPPC	0.35	0.68	0.01	-0.04	0.69	-0.51	1.00					
PCAP	0.34	0.77	0.05	-0.02	0.67	-0.43	0.82	1.00				
PIMPC	0.03	0.05	-0.54	0.75	0.87	0.21	0.43	0.40	1.00			
PIMPDU	-0.11	-0.16	-0.75	0.77	0.74	0.24	0.29	0.15	0.78	1.00		
SGDPPC	0.16	0.17	-0.56	0.69	0.98	0.03	0.54	0.55	0.90	0.79	1.00	

Source: Computed

Granger causality among petroleum import, output, and carbon emission

The results of the multivariate linear and nonlinear Granger causality tests conducted among aggregate and sectoral output, petroleum import, and carbon emission (in per capita terms) are reported in Table 5. For both Nigeria and South Africa, there is no evidence of nonlinear Granger causality among the variables for almost all the models. At the aggregate level, results for Nigeria show that linear and nonlinear causality runs from GDP to carbon emissions and petroleum import without any feedback. This suggests that economic growth can cause the petroleum import-carbon emission link in Nigeria, which may be traced to the ability to promote the use of carbon-reducing techniques in the face of continual consumption of petroleum import. Further, unidirectional linear causality moves from carbon emissions to petroleum import and GDP, and from petroleum import to carbon emissions and GDP. Thus, concerns for the environment can alter the relationship between income growth and consumption of imported petroleum products, just as changes in petroleum imports have serious implication for growth-environment nexus in the country. At sectoral level, linear Granger causality is observed to run jointly from carbon emission and petroleum import per capita to each of Nigeria's agricultural and service output while the reverse causality is only valid for the case of manufacturing output. This indicates that, in the presence of petroleum import, concern for the environment has major influence on agricultural and service output performance, while manufacturing output yields a major influence on the relationship between petroleum import and the environment. However, for the agricultural and manufacturing sectors, output and petroleum import jointly Granger cause carbon emissions. For service sector, feedback effect is found. Moreover, carbon emission combined with each of agricultural and service output to produce unidirectional causality running to petroleum import. While this effect is not observed for the manufacturing sector, the reverse causality is evident. There is therefore evidence of different multivariate links at sectoral levels producing important impacts on each variable in Nigeria.

In the case of South Africa, results from aggregate analysis could only establish linear Granger causality jointly running from carbon emissions and GDP to petroleum import, with

Variables	Convention	nal unit root test	Unit root wi	th structural break(s)	Nonlinear	unit root tests	Final deci	ision
	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa
CO2PC	I(1)	I(1)	I(0)	I(0)	I(1)	I(1)	I(1)	I(1)
EXR	I(1)	I(0)	I(1)	I(1)	I(1)	I(0)	I(1)	I(0)
FDEV	I(1)	I(1)	I(0)	I(0)	I(0)	I(1)	I(0)	I(1)
GDPPC	I(1)	I(1)	I(0)	I(0)	I(1)	I(0)	I(1)	I(0)
AGDPPC	I(1)	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
MGDPPC	I(1)	I(1)	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
SGDPPC	I(1)	I(1)	I(0)	I(1)	I(0)	I(0)	I(0)	I(1)
HCAP	I(1)	I(1)	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
PCAP	I(0)	I(1)	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
PIMPC	I(1)	I(1)	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
ТО	I(1)	-	I(1)	-	I(0)	_	I(1)	_

Table 4 Overall decision across the conventional, unit root with structural break(s) and nonlinear unit root

similar finding observed in the service sector. In the agricultural sector, unidirectional linear Granger causality runs from output to carbon emissions and petroleum import and from carbon emissions to output and petroleum import. Thus, agricultural output performance is important for the link between consumption of imported petroleum products and environmental pollution, which may in turn dictate the relationship between agricultural output and petroleum import. There is also linear Granger causality running jointly from carbon emission and agricultural output to petroleum import. For the manufacturing sector, carbon emissions Granger causes output and petroleum import, while this import causes both output and carbon emissions. Thus, environmental targets have important simultaneous consideration for manufacturing output and petroleum imports, which may mediate the link between manufacturing output and pollution.

The foregoing results of the multivariate causality method provide evidence of linear, direct, indirect (mediating and joint), and simultaneous effects of the variables on one another. In order to properly capture these effects, there is a need to set up a simultaneous equation model which will not only clearly show the effects but will also resolve the potential endogeneity issue and provide empirical estimates that will permit the calculation of net effects of growth-emission nexus and net effect of growth-energy import link that will serve as indicator of the levels of attainment of sustainable growth in South Africa and Nigeria. These estimates would be useful for policy analysis.

Results from simultaneous equations model (SEM)

The complete link, as well as magnitude of impact, among energy import, economic growth, and carbon emissions is further investigated using simultaneous equation model in the presence of a number of control variables. The article estimates both the three-stage least square (3SLS) and the system generalized method of moments (sys-GMM) with similar results, both in sign and magnitude. This possibility is confirmed in the literature as multiple-equation GMM tends to reduce to full-information instrumental variable (IV) efficient estimator when conditional homoscedasticity is present. It ultimately reduces to 3SLS when all the equations have identical set of IVs (Hayashi 2000; and Adewuyi and Awodumi 2017b). The preference for 3SLS over the other instrumental variable (IV) approaches (2SLS and GMM) is based on the idea that it could correct for both endogeneity problem and contemporaneous correlation of the error terms across equations (Hayashi 2000; and Adewuyi and Awodumi 2017b). Hansen tests, which are used to verify the validity of the instruments employed, are insignificant for all the models, suggesting that the instruments are valid.

The full results from the 3SLS are provided in Appendix 3 (Tables 16, 17, 18, and 19) for aggregate and sectoral (agricultural, manufacturing, and service) output. For the purpose of clarity, summarized 3SLS results are used in the aggregate and sectoral analyses below with figures. Further analysis is conducted by examining the achievement of sustainable economic growth (SEG) in South Africa and Nigeria via the validation of the proposition that SEG is achieved when the contribution of income per capita to the improvement in environmental quality and energy efficiency outweighs the relative contribution of both petroleum import and environment to income growth (Zuo and Ai 2011; Hu et al. 2011; Omer 2008;). This means the attainment of clean production (or low carbon development) and energy efficiency.

Petroleum import, aggregate output, and carbon emissions in South Africa and Nigeria

Based on the 3SLS estimates of the SEM, the nexus among petroleum import, aggregate output, and carbon emission is

Table 5 Results of multivar	iate linear and nonlinear Grar	nger causality test fo	or Nigeria and Sou	tth Africa		
Variables	Direction of causality	Bound test F-statistics	Framework	Linear causality chi square	Nonlinear causality F-statistics	Results
Nigeria						
CO2, RIMP and GDP	CO2, PIMP→GDP GDP → CO2, PIMP	0.107	VAR (1)	0.023 3.683**	0.059 3.228***	Linear and nonlinear causality from GDP to CO2 and PIMP
	CO2 → PIMP, GDP PIMP, GDP → CO2	7.689**	VEC (3)	6.816*** 4.057	0093 0.182	Linear causality runs from CO2 to PIMP and GDP No nonlinear causality
	PIMP→GDP, CO2 CO2, GDP → PIMP	3.069	VAR (3)	7.639** 5.059	0.315 0.157	Linear causality from PIMP to CO2 and GDP No nonlinear causality
CO2, PIMP and AGDP	CO2, PIMP→AGDP AGDP→CO2, PIMP	0.234	VAR (3)	15.107* 2.129	0.001 0.105	Linear causality from CO2 and PIMP to AGDP No linear and nonlinear causality
	CO2 → PIMP, AGDP PIMP, AGDP→CO2	3.728	VAR (3)	2.743 6.554***	0.075 0.432	Linear causality from PIMP and AGDP→CO2 No nonlinear causality
	PIMP→AGDP, CO2 CO2, AGDP→PIMP	3.034	VAR (4)	4.072 11.070**	0.107 0.738	Linear causality from CO2 and AGDP to PIMP No linear and nonlinear causality
CO2, PIMP and MGDP	CO2, PIMP→MGDP MGDP→CO2, PIMP	0.258	VAR (1)	0.373 3.746**	$0.004 \\ 0.098$	Linear causality from MGDP to CO2 and PIMP No nonlinear causality
	CO2 → PIMP, MGDP PIMP, MGDP →CO2,	6.909**	VEC (1)	0.694 2.676***	0.029 0.032	Linear causality from PIMP and MGDP to CO2 No nonlinear causality
	PIMP→MGDP, CO2 CO2, MGDP→PIMP	3.412	VAR (3)	10.776* 2.781	0.112 0.755	Linear causality from PIMP to MGDP and CO2 No nonlinear causality
CO2, PIMP and SGDP	CO2, PIMP→SGDP SGDP→CO2, PIMP	1.794	VAR (5)	16.846* 2.801	0.317 1.089	Linear causality from CO2 and PIMP to SGDP No nonlinear causality
	CO2 → PIMP, SGDP PIMP, SGDP →CO2,	3.294	VAR (3)	6.910*** 6.771***	$0.084 \\ 0.635$	Linear causality from CO2 to PIMP and SGDP and reverse No nonlinear causality
	PIMP→SGDP, CO2 CO2, SGDP→PIMP	2.947	VAR (3)	5.230 7.412**	0.127 0.137	Linear causality from CO2 and SGDP to PIMP No nonlinear causality
South Africa						
CO2, RIMP and GDP	CO2, PIMP→GDP GDP → CO2, PIMP	0.917	VAR (2)	1.013 1.369	0.076 0.751	No linear and nonlinear causality
	CO2 → PIMP, GDP PIMP, GDP → CO2	3.423	VAR (2)	0.176 0.200	0.801 0.105	No linear and nonlinear causality
	PIMP→GDP, CO2 CO2, GDP → PIMP	3.252	VAR (3)	0.379 10.177*	$0.253 \\ 0.308$	Linear causality from CO2 and SGDP to PIMP No nonlinear causality
CO2, PIMP and AGDP	CO2, PIMP→AGDP AGDP→CO2, PIMP	10.804^{*}	VEC (3)	1.344 9.193**	$0.192 \\ 0.439$	Linear causality from AGDP to PIMP and CO2 No linear and nonlinear causality
	CO2 → PIMP, AGDP PIMP, AGDP→CO2	16.154*	VEC (3)	12.165* 3.194	0.027 0.183	Linear causality from CO2 to RIMP and AGDP No nonlinear causality
	PIMP→AGDP, CO2 CO2, AGDP→PIMP	17.999*	VEC (2)	0.306 5.872**	$0.181 \\ 0.193$	Linear causality from CO2 and AGDP to PIMP No linear and nonlinear causality

Table 5 (continued)						
Variables	Direction of causality	Bound test F-statistics	Framework	Linear causality chi square	Nonlinear causality F-statistics	Results
CO2, PIMP and MGDP	CO2, PIMP→GDP MGDP→CO2, PIMP	2.405	VEC (3)	1.988 0.748	0.175 0.214	No linear and nonlinear causality
	CO2 → PIMP, MGDP PIMP, MGDP →CO2,	6.866**	VEC (1)	5.040***0.459	0.106 0.189	Linear causality from CO2 to PIMP and MGDP No nonlinear causality
	PIMP→MGDP, CO2 CO2, MGDP→PIMP	8.653**	VEC (5)	10.231^{***} 3.090	0.407 0.181	Linear causality from PIMP to CO2 and MGDP No nonlinear causality
CO2, PIMP and SGDP	CO2, PIMP→SGDP SGDP→CO2, PIMP	2.166	VAR (2)	1.131 1.110	0.231 0.523	No linear and nonlinear causality
	CO2 → PIMP, SGDP PIMP, SGDP →CO2,	2.284	VAR (2)	0.025 0.279	0.867 0.110	No linear and nonlinear causality
	PIMP→SGDP, CO2 CO2, SGDP→PIMP	2.918	VAR (3)	0.104 8.479*	0.211 0.300	Linear causality from CO2 and SGDP to PIMP No nonlinear causality
The linear Granger causality t	est is based on a VAR model	for the logged raw	data, while the no	onlinear causality is bas	ed on residuals fro	om a VAR model for he logged raw data

, and * represent significance levels at 1%, 5%, and 10%, respectively. Standard errors are in parentheses. CO2 = carbon emission per capita; PIMP = petroleum import per capita; GDP = GDP per

capita; AGDP = agricultural GDP; MGDP = manufacturing GDP; SGDP = service GDI

presented in Fig. 1. It can be seen that, with and without accounting for structural break in petroleum import, there are incomplete links among the variables. Therefore, growth-emission nexus occurs in both South Africa and Nigeria, while petroleum import-emission nexus exists only in Nigeria. However, petroleum import-growth nexus is observed only without accounting for structural break in both countries.

When structural break in petroleum import is considered, growth and emission have significant negative effect on each other in both South Africa and Nigeria (Table 6). This means that 1% rise in growth in South Africa and Nigeria reduces emission by 2.005% and 2.930%, respectively, while 1% increase in emission also reduces growth by 0.491% and 0.298%, respectively. This yielded the net effects of 1.514% and 2.632% reduction in emission, respectively, supporting the low carbon development or clean production aspect of the sustainable development (Zuo and Ai 2011). This feedback (bidirectional) hypothesis is confirmed by Apergis and Payne (2014) for the case of 7 Central American countries. Further, with and without accounting for structural break, petroleum and emission have significant negative effect on each other in Nigeria, a bidirectional relationship that is also supported by Al-Mulali (2011) and Azam (2019). Thus, 1% rise in petroleum import reduces emission by 0.180%, while 1% increase in emission also reduces petroleum import by 4.115%, and the net effect is 3.935% reductions in petroleum import. For South Africa, only emission has significant positive effect on petroleum import in the absence of structural break. This means that 1% rise in emission increases petroleum import by 1.820%. Also, in the absence of structural break, growth and petroleum import have significant positive effect on each other in both South Africa and Nigeria. Thus, 1% rise in petroleum import raises output in South Africa and Nigeria by 0.110% and 0.142%, respectively, while 1% rise in output also increases petroleum import by 7.302% and 5.443% respectively, and the net effects are 7.192% and 5.301% increases in petroleum import, respectively. These results indicate the existence of feedback hypothesis between petroleum import and economic growth, in line with Saboori and Sulaiman (2013), Lim et al. (2014), Park and Yoo (2014), and Azam (2019).

Against the above analysis, the proposition that sustainable growth can be attained with clean production or low carbon development (arising from the net effect of growth-emission nexus) and energy efficiency (stemming from the net effect of growth-petroleum import nexus) cannot be established for South Africa and Nigeria irrespective of the assumption of structural break in the analysis. Specifically, although the net effect of growth-emission nexus (which is clean production) is established for both countries, considering the structural break, the net effect of growth-petroleum import nexus cannot be estimated for them due to statistically insignificant results.



Fig. 1 Aggregate economic growth, petroleum import, and CO_2 emission in Nigeria and South Africa. Source: Author; from regression results: NG, Nigeria; SA, South Africa; WB, model with structural break; WTB, model without structural break

Also, although the net effect of growth-petroleum import nexus is established, without considering the structural break, the net effect of growth-emission nexus cannot be estimated for the countries due to statistically insignificant results. Besides, the estimated net effect of growth-petroleum import nexus is positive and high in terms of energy use, which means energy inefficiency.

Petroleum import, agricultural output, and carbon emissions in South Africa and Nigeria

The relationship among petroleum import, agricultural sector output, and carbon emission is shown in Fig. 2 which is derived from the results of the 3SLS. There are overwhelming partial (bivariate) links between the variables. Without accounting for a structural break, growth-emission nexus exists in both South Africa and Nigeria (also hold accounting for structural break in Nigeria's case). Further, petroleum import-emission nexus occurs in Nigeria without accounting for structural break, while it exists in South Africa accounting for a structural break. However, growth-petroleum import nexus occurs only in Nigeria without accounting for a structural break. Complete output-energy-emission link exist for Nigeria when a structural break is not considered.

Without accounting for structural break, growth and emission generate significant negative effect on each other in South Africa but reveal significant positive effect in Nigeria (Table 6). A significant negative bidirectional impact is exhibited accounting for structural break only in Nigeria's case. Thus, 1% rise in growth engenders a fall in emission by 0.952%, while similar increase in emission leads to 1.039% reduction in growth in South Africa. Also, 1% improvement in economic growth in Nigeria raises (reduces) the level of carbon emission by 0.338% (1.093%) as a similar increase in emission contributes 2.882% (0.673%) to the increase (reduction) in the economic growth without (considering) the influence of structural break (Table 6). These bidirectional links between growth and emission were confirmed by Adewuyi and Awodumi (2017b) in 7 West African countries, including Nigeria. In Nigeria, the net effects are 0.087% fall in emission and 2.544% rise in output with and without structural break, respectively, and this further confirms the proposition of Zuo and Ai (2011). In South Africa, the net effect is 0.141% reduction in output.

Further, petroleum import and emission produce negative effect on each other in Nigeria (without accounting for a structural break), while they also generate similar effect in South Africa (accounting for structural break). Therefore, 1% increase in petroleum import leads to 0.261% fall in emission in Nigeria, while a similar rise in emission results in 3.846% fall in petroleum import and the net effect is 3.585% fall in Nigeria's petroleum import. In the case of South Africa, 1% increase in petroleum import leads to 0.041% fall in emission, while a similar rise in emission results in 4.44% falls in petroleum import and the net effect is 4.399% fall in South Africa's petroleum import. These findings are supported by Al-Mulali (2011) and Shafiei and Salim (2014) for MENA and OECD countries, respectively.

Moreover, growth and petroleum import exert positive impact on each other only in Nigeria (without accounting for structural break). Thus, 1% rise in growth leads to 1.30% increase in petroleum import while a similar rise in petroleum import raises growth by 0.771% in Nigeria yielding a net effect of 0.529% increase in Nigeria's petroleum import. While the results for the case of South Africa supports the growth hypothesis similar to Awodumi and Adewuyi (2020) for Angola, Egypt, and Nigeria, the findings for Nigeria follow the feedback hypothesis in the energy-growth literature as confirmed by Ziramba (2009) in the author's hypothesis between crude oil and economic growth in South Africa. Thus, there is no evidence of complete links among petroleum import, growth, and carbon emission in the agricultural sector in both countries.

The foregoing analysis of the agricultural sector shows evidence of positive net effect on the environment (with or without a structural break) and increasing net effect on petroleum import (without a structural break) in Nigeria. In South Africa, there is an evidence of negative net effect on output without accounting for a structural break. Thus, the sustainable growth proposition (existence of clean production or low carbon development and energy efficiency) in the agricultural sector could not be established in both countries. In particular, although in Nigeria, the net effect of growth-emission nexus suggests a clean production, the net effect of growth-petroleum import nexus suggests energy inefficiency. In the case of South Africa, the net effect of growth-emission nexus does not support a clean production while the net effect of growth-petroleum import nexus cannot be determined as estimates are statistically insignificant.

Country	$\rm CO2 \rightarrow \rm GDP$	$GDP \rightarrow CO2$	CO2↔GDP	PIMP→GDP	$\mathrm{GDP} \to \mathrm{PIMP}$	PIMP↔GDP	$CO2 \rightarrow PIMP$	PIMP→C02	CO2⇔PIMP
Aggregate economy									
Nigeria with structural break	✔(-0.298)	✓(-2.930)	>	√ -0.063	Х	X	✓(-4.115)	✔(-0.180)	>
Nigeria	✔(0.272)	х	Х	✔(0.142)	✓(5.443)	~	✓(−1.675)	✔(-0.162)	>
No structural break South Africa	√ (-0.491)	✓(-2.005)	`	×	X	X	X	×	×
With structural break						1			1
South Africa No structural break	Х	✔(-1.614)	X	✓(0.110)	✓(7.302)	>	✔(1.820)	Х	Х
Agricultural sector									
Nigeria	✔(-0.673)	✔(-1.093)	>	Х	Х	Х	Х	Х	Х
w IUI SUUCIUIAI UICAN Nigeria	✔(2.882)	✔(0.338)	>	✓(0.771)	✓(1.304)	~	✓(-3.846)	✔(-0.261)	>
No structural break									
South Africa	Х	Х	Х	✔(0.179)	Х	Х	✔(-4.44)	✔(-0.041)	>
W IUI SUUCIUIAL DICAR South Africa	✓(-1.039)	✓(-0.952)	>	(−0.097)	Х	X	Х	√ (0.089)	X
No structural break Manufacturing sector		r.							
Nigeria	√ (2.238)	√ (0.417)	`	✔(0.573)	✔(1.478)	>	✓(-3.809)	✔(-0.258)	>
With structural break Nigeria	J(2 155)	(920)	``	J(0 555)	7(1 564)	7	J(-3 741)	7(-0.261)	``
No structural break	(001:2)		•			•			•
South Africa	✔(0.664)	√ (1.239)	>	✔(0.103)	V (6.244)	~	✓(-6.521)	✓(-0.158)	`
With Structural Dreak South Africa	(632)	(0801)7	,	X	116 077)	X	X	X	X
No structural break Service sector			•	<		4	<	4	\$
Nigeria With etmotural break	✔(0.752)	Х	Х	X	X	Х	✓(-3.674)	Х	Х
Nigeria	Х	Х	Х	✔(0.452)	X	Х	✔(-3.392)	✔(-0.66)	`
No structural break South Africa	V (0,300)	J(2.851)	`	×	×	×	×	1 (-0.084)	×
With Structural Break				1	1	4	1		4
South Africa No structural break	✔(0.369)	✔(2.340)	>	√ (0.019)	×	×	×	✔(-0.042)	×
Source: Extracted from Table 10 i	n the Appendix. A	GG=aggregate mod	lel; AGR = agricul	tural sector; MAN :	= manufacturing sec	tor; SER = service (sector; CO2 = carbo	n emission per capi	ta; GDP = GDP
per capita; $PIMP = petroleum Im_{i}$	oort per capita								



Fig. 2 Agriculture sector growth, petroleum import, and CO2 emission in Nigeria and South Africa. Source: Author; from regression results: NG, Nigeria; SA, South Africa; WB, model with dummy; WTB, model without structural break

Petroleum import, manufacturing output, and carbon emissions in South Africa and Nigeria

Following the results of the 3SLS, the petroleum importmanufacturing output-emission nexus is shown in Fig. 3. There is a complete link among the variables in both countries when the structural break in petroleum import is taken into consideration. However, while a complete link still exists among the variables in the absence of structural break in the case of Nigeria, it does not occur in South Africa. Thus, there is only a growth-emission link in South Africa in the absence of structural break.

For the manufacturing sector, with and without considering a structural break in petroleum import (estimates are very close), emission and growth exert positive effect on each other (2.238% and 0.417%) in Nigeria (Table 6). A similar positive effect also occurs between petroleum import and growth (0.573% and 1.478%). However, emission and petroleum import have negative effect on each other (-3.859% and -



Fig. 3 Manufacturing sector growth, petroleum import, and CO2 emission in Nigeria and South Africa. Source: Author; from regression results: NG, Nigeria; SA, South Africa; WB, model with dummy; WTB, model without structural break

0.358%). Thus, while the net effect of emission-growth nexus is 1.821% increase in output, the net effect of petroleum import-growth nexus is 0.905% increase in petroleum import and that of emission-petroleum import nexus is 2.551% fall in petroleum import. Overall, while the growth rises by 1.821%, that of petroleum import declines by 1.646%. These results appear not compatible with sustainable development as suggested by Zuo and Ai (2011) that the contribution of growth to energy efficiency and environment is lower than those contributed by energy input (import) and environment. This is more plausible following the call for improved sustainability and energy conservation of industrial production by Safarzadeh et al. (2020).

Recognizing the structural break, emission and growth exert positive effect on each other (0.664% and 1.239%) in South Africa. A similar positive effect also occurs between petroleum import and growth (0.103% and 6.244%). However, emission and petroleum import have negative effect on each other (-6.521% and -0.158%). Thus, while the net effect of emission-growth nexus is 0.575% increase in emission, the net effect of petroleum import-growth nexus is 5.141% increase in petroleum import and that of emission-petroleum import nexus is 6.363 fall in petroleum import. Overall, while emission rises by 0.575%, petroleum import growth declines by 1.222%. In the absence of structural break, emission and growth generate positive effect on each other. Thus, 1% rise in emission raises growth by 0.632% while similar increase in growth raises emission by 1.089%. The net effect rises in emission by 0.457%. Irrespective of the influence of structural break, complete feedback hypothesis is supported among all petroleum import, carbon emission, and economic growth in both countries (except South Africa without structural break) in the manufacturing sector.

The implications of the above analysis for sustainable growth are evident. The sustainable growth proposition is not established in both countries in the manufacturing sector regardless of the existence of structural break in petroleum import. Specifically, with or without the structural break, the emission-growth nexus in Nigeria generates a positive net effect on output contributing to a clean production or low carbon development, but the petroleum import-growth nexus yields increasing net effect on petroleum import producing energy inefficiency. For South Africa, the net effects of emission-growth nexus and petroleum import-growth nexus raise emission and petroleum import, respectively, in the presence of the structural break. For the same country, similar increasing net effect on carbon emission is found (for the emission-growth nexus), though an insignificant petroleum import-growth nexus is revealed when the structural break is not considered. Thus, in South Africa, there is no evidence of clean production and energy efficiency in the manufacturing sector.

Petroleum import, service sector output, and carbon emissions in South Africa and Nigeria

Based on the estimates from the 3SLS, the petroleum importservice output-emission links are exhibited in Fig. 4. With and without structural break in petroleum import, there are partial links among the variables in both countries. With respect to South Africa, emission-growth nexus occurs with and without considering the structural break. However, for Nigeria, emission-petroleum import nexus exists without recognizing the structural break.

With and without structural break in petroleum import, emission and growth produce positive effect on each other (0.300% and 2.851%) in the service sector in the case of South Africa (Table 6). The net effect rises in emission by 2.551%. These results are in line with Apergis and Payne (2014), Adewuyi and Awodumi (2017b), and Huo et al. (2020) for 7 Central American countries, 7 West African countries, and 30 provinces of China, respectively. However, for Nigeria, emission and petroleum import exert negative effect on each other (-3.392% and -0.660%) when the structural break is not recognized. The net effect is fall in petroleum import by 2.732%. This bidirectional effect is reported by Al-Mulali (2011) for MENA countries. In both countries, growth hypothesis is confirmed between petroleum import and economic growth in the service sector, as confirmed by Awodumi and Adewuyi (2020) where petroleum consumption is found to have a significant effect on the economic growth in Angola, Egypt, and Nigeria.

The implication of the above analysis is that sustainable growth, which combines clean production and energy efficiency, cannot be confirmed in the service sector for South Africa and Nigeria, regardless of the structural break. In South Africa, the negative net effect of the emission-growth nexus

Fig. 5 Links among economic growth, petroleum import, and CO2 emission in Nigeria and South Africa (with structural break dummy). Source: Author; from regression results: NG, Nigeria; SA, South Africa; AGG, aggregate GDP; AGR, agricultural GDP; MAN, manufacturing GDP; SER, service GDP

on the environment could not support clean production while the growth-petroleum import nexus could not be determined. Also, estimates could not provide support for both the emission-growth nexus and growth-petroleum import nexus in Nigeria.

The summary of the interactions among GDP per capita, petroleum import per capita, and carbon emission per capita are further presented in Figs. 5 and 6 (based on Table 6) for model with and without petroleum import structural break, respectively. Complete significant interactive effects are found among these variables in the Nigerian manufacturing sector irrespective of the effect of the structural break in petroleum import. However, in South Africa, this link is only confirmed in this sector when such break is accounted for. This is indicative of the heavy dependence on imported

Fig. 4 Service sector growth, petroleum Import and CO2 Emission in Nigeria and South Africa. Source: Author; from regression results: NG, Nigeria; SA, South Africa; WB, model with dummy; WTB, model without structural break

Fig. 6 Links among economic growth, petroleum import, and CO2 emission in Nigeria and South Africa (without structural break dummy). Source: Author; from regression results: Note: NG, Nigeria; SA, South Africa; AGG, aggregate GDP; AGR, agricultural GDP; MAN, manufacturing GDP; SER, service GDP

refined petroleum products for manufacturing activities in these countries. While the consumption of such product contributes immensely to increase in manufacturing output, the carbon emission implication is enormous. In the same vein, increase in economic activities in this sector yields higher output that encourages further consumption of imported refined petroleum products with similar emission implication, which in turn affects the manufacturing productivity through the state of health of the labor force. This nexus is also confirmed for Nigeria for the aggregate and agricultural output when the structural break is neglected. Generally, feedback and partial impacts are common among the variables in both countries, though these effects vary with and without accounting for structural break in petroleum import.

The robustness of the above analyzed results obtained from the 3-stage least squares (3SLS) estimator is checked by estimating the simultaneous equation model with system generalized method of moments (sys-GMM). As stated earlier, the results obtained are very similar in signs and magnitudes of the coefficients of the explanatory variables. The similarity in the results is highlighted in the econometrics literature given that multiple-equation GMM is likely to collapse to fullinformation instrumental variable (IV) efficient estimator with the presence of conditional homoscedasticity. System-GMM eventually diminishes to 3SLS since the three equations estimated in this study have identical set of IVs (Hayashi 2000; and Adewuyi and Awodumi 2017b). However, the 2SLS could not produce better results. Given that the results are robust with two different estimators, we proceed to carry out further analysis using threshold regression that could provide a framework for policy analysis.

Threshold analysis of energy (petroleum) importcarbon emission-growth nexus

The study conducts further analysis to determine the optimal level of per capita petroleum import, carbon emission, and economic growth that is necessary to spur sustainable economic growth. The optimal levels of per capita petroleum import below (regime 1) or above (regime 2) for which the additional import of this energy mediates the link between economic growth and carbon emission are reported in Table 7.⁸ The threshold value of petroleum import per capita for the effect of GDP per capita on carbon emission per capita is 3.42 for aggregate analysis for the case of Nigeria. For the same country, the results show that below the threshold level of petroleum import per capita, aggregate and service GDP per capita do not exert significant effect on carbon emission per capita has an escalating effect on the carbon emission per capita.

However, over this regime, the agricultural GDP per capita has a reducing effect on the carbon emission per capita. Above the threshold, the aggregate and all sectoral output per capita contribute significantly to the reduction in the carbon emission per capita. In South Africa, the threshold level of petroleum import per capita varies across sectors. Nevertheless, both the aggregate and manufacturing output per capita raise the level of CO2 per capita significantly in both regimes (above and below the threshold). However, while the agricultural output per capita has a significant positive influence on the carbon emission per capita only in regime 2, the service output per capita has a similar effect only in regime 1.

The threshold values of the petroleum import per capita for which it mediates the effect of carbon emission per capita on GDP per capita is the same as the aggregate for all sectors considered in Nigeria but they vary across sectors in South Africa.⁹ In Nigeria, the influence of carbon emission per capita on aggregate, as well as manufacturing output per capita, is significantly positive and negative below and above the threshold (3.88 and 3.87, respectively). For other sectors (Agriculture and Service), carbon emission per capita has a reducing effect on the output per capita in both regimes. In South Africa, the impact of the carbon emission per capita on the aggregate and service output per capita is significant (positive) only in regime 1 while a similar significant impact is observed only in regime 2 for the agricultural sector. For the same country, this emission per capita exerts a significant positive influence on the manufacturing output per capita in both regimes.

Generally, in Nigeria, GDP per capita has a reducing effect on the carbon emission per capita above the threshold level at aggregate and sectoral levels. Thus, keeping petroleum import per capita above the respective threshold enhances environmental quality as production activities increase. However, the CO2 emission can only induce increased GDP per capita when the petroleum import is below the threshold level. There is therefore a policy dilemma for the country, which implies that the government must strive to achieve optimal utilization of energy import so as to attain a low carbon development via increasing the application of carbon-reducing production techniques as income increases.

In South Africa, a significant positive feedback effect is found between the carbon emissions per capita and output per capita in both regimes for the manufacturing sector. Thus, keeping petroleum import below the threshold value may manage the adverse effect of higher output on the environment. However, results vary across both threshold regimes for other sectors. This may reflect the varied level of utilization and sensitivity of the sectors to petroleum requirements. Consequently, in the agricultural and service sectors, keeping

⁸ The relevant threshold equation is given as: $[CO2PC = f(GDPPC; I(PIMPC < \gamma \text{ or } \geq \gamma)]$

⁹ The relevant threshold equation is given as [GDPPC = f (CO2PC; *PIMPPC* $< \gamma \text{ or } \geq \gamma$)]

 Table 7
 Threshold results for the relationship between CO2 emission per capita and GDP per capita: threshold variable = petroleum import per capita

Country	Sector	PIMPC	Bootstrap	Regime 1 ($\alpha_{it} \leq \gamma$	()	Regime 2 ($\alpha_{it} \ge$	(γ)
		threshold value	LM test	Intercept (s.e.)	GDPPC/CO2PC (s.e.)	Intercept (s.e.)	GDPPC/CO2PC (s.e.)
Dependent varia	able, CO2PC; inde	ependent variable,	GDPPC				
Nigeria	Aggregate	3.42	19.56*	-10.61(13.33)	2.32(1.86)	9.66(0.64)*	-0.42(0.08)*
	Agriculture	3.42	17.31*	24.42(7.37)*	-3.34(-1.33)**	8.27(0.24)*	-0.30(0.04)*
	Manufacturing	3.42	18.63*	1.35(1.68)	1.01(0.37)**	7.98(0.27)*	-0.30(0.06)*
	Service	3.42	18.17*	0.85(15.15)	0.84(2.47)	8.41(0.23)*	-0.29(0.03)*
South Africa	Aggregate	3.25	9.45*	-4.01(1.64)**	1.50(0.19)*	6.87(1.04)*	0.25(0.12)**
	Agriculture	2.91	3.37	7.29(1.09)*	0.36(0.21)	7.67(0.48)*	0.28(0.09)*
	Manufacturing	3.21	11.69*	2.72(0.79)*	0.94(0.15)*	6.02(0.83)*	0.45(0.12)*
	Service	3.31	8.23**	-6.08(1.70)*	1.86(0.21)*	7.50(0.90)*	0.20(0.11)
Dependent varia	able, GDPPC; ind	ependent variable,	CO2PC				
Nigeria	Aggregate	3.88	16.58*	6.35(0.28)*	0.13(0.05)**	15.45(1.56)*	-1.22(0.24)*
	Agriculture	3.87	22.62*	6.63(7.06)	-0.19(0.04)*	14.87(2.36)*	-1.35(0.37)*
	Manufacturing	3.87	9.90*	3.45(0.59)*	0.19(0.10)	16.36(3.08)*	-1.77(0.47)*
	Service	3.87	18.24*	6.79(0.22)*	-0.11(0.04)**	22.00(2.50)*	-2.36(0.38)*
South Africa	Aggregate	3.65	21.36*	5.28(0.96)*	0.38(0.10)*	8.36(2.33)*	0.06(0.25)
	Agriculture	2.73	3.29	2.08(2.71)	0.33(0.30)	-1.45(2.22)	0.72(0.24)*
	Manufacturing	4.07	10.74*	0.61(1.40)	0.68(0.15)*	0.56(1.16)	0.69(0.13)*
_	Service	3.65	21.92*	6.25(0.88)*	0.21(0.10)**	8.55(3.36)**	-0.02(0.37)

Source: Author

s.e. = standard error; CO2PC = carbon emission per capita; PIMPC = petroleum import per capita. Levels of significance at 1%, 5%, and 10% are represented as *, **, and ***, respectively

petroleum import below and above the threshold (respectively) is important, as such neither raise the carbon emission per capita nor reduce the output per capita significantly. On the aggregate, although maintaining the petroleum import beyond the threshold may increase the CO2 emission per capita which is associated with high output per capita, such emission exhibits negligible reverse impact on output per capita.

The optimal levels of per capita carbon emission below or above which an increase in the level of CO2 emission per capita has implication for the link between GDP per capita and petroleum import per capita are presented in Table 8. For the case of Nigeria, the threshold value of carbon emission per capita remains the same (6.46) for aggregate and sectoral analysis. Above or below this threshold, petroleum import per capita exerted positive effect on aggregate and agricultural output per capita. However, the impact of this import on manufacturing GDP per capita and service GDP per capita is significantly positive only below and above the threshold levels, respectively. In South Africa, the aggregate and service output per capita are positively affected by the levels of petroleum import per capita in both regimes, but the impact of this import exerts a similar effect on the manufacturing output only below the threshold. For agricultural sector, no significant effect of this import on output is observed in both regimes.

The threshold level of carbon emission per capita remains the same (6.46) across sectors in Nigeria but varies in South Africa. For the effect of per capita output on petroleum import per capita, the results for Nigeria show that aggregate, agricultural, and service output increase per capita import of petroleum products below or above carbon emission threshold levels. The influence of the manufacturing output per capita on petroleum import per capita is only significant (positive) in regime 1. In South Africa, the per capita output has a significant (positive) impact on the petroleum import per capita only in regime 1 for the aggregate, manufacturing, and service sectors. The effect of the agricultural output per capita on this import is negligible either below or above the threshold level of the CO2 emission per capita.

These results suggest that the increase in petroleum import largely promotes in the economic growth both at aggregate and sectoral levels irrespective of the level of carbon emissions in both countries. However, keeping carbon emission below its threshold levels enhances the positive influence of economic growth on petroleum import per capita in both countries. Interestingly, as carbon emission rises above the threshold in South Africa, the effect of growth on petroleum import is neutralized at aggregate and sectoral levels, respectively.

 Table 8
 Threshold results for the relationship between petroleum import per capita and GDP per capita: threshold CO2 emission per capita

Country	Sector	CO2PC	Bootstrap	Regime 1 ($\alpha_{it} \leq \gamma$	γ)	Regime 2 ($\alpha_{it} \ge$	γ)
		threshold value	LM test	Intercept (s.e.)	PIMPC/GDPPC (s.e.)	Intercept (s.e.)	PIMPC/GDPPC (s.e.)
Dependent varia	ble, GDPPC; inde	pendent variable, PI	MPC				
Nigeria	Aggregate	6.46	12.10*	6.23(0.15)*	0.36(0.04)*	5.84(0.59)*	0.37(0.15)*
e	Agriculture	6.46	13.91*	4.28(0.22)*	0.47(0.06)*	0.94(0.55)	1.20(0.15)*
	Manufacturing	6.46	11.13*	3.59(0.23)*	0.38(0.07)*	4.78(0.16)*	-0.03(0.04)
	Service	6.46	15.55*	4.71(0.24)*	0.55(0.68)	2.89(0.63)*	0.84(0.17)*
South Africa	Aggregate	9.08	8.21*	8.65(0.02)*	0.02(0.01)**	8.44(0.05)*	0.11(0.01)*
	Agriculture	9.13	5.93	5.03(0.11)*	0.03(0.03)	5.34(0.12)*	-0.04(0.03)
	Manufacturing	9.17	12.15*	6.60(0.04)*	0.06(0.01)*	6.95(0.10)*	-0.01(0.03)
	Service	9.08	7.45**	8.08(0.03)*	0.03(0.01)*	7.72(0.06)*	0.16(0.02)*
Dependent varia	ble, PIMPC; indep	oendent variable, GI	OPPC				
Nigeria	Aggregate	6.46	18.94*	-12.24(1.96)*	2.10(0.25)*	-1.73(1.46)	0.78(0.20)*
C	Agriculture	6.46	19.27*	-6.07(1.11)*	1.61(0.18)*	0.51(0.38)	0.61(0.08)*
	Manufacturing	6.46	14.93*	-4.54(1.22)*	1.62(0.24)*	4.05(1.74)*	-0.02(0.37)
	Service	6.46	18.70*	-5.62(1.12)*	1.37(0.16)*	-0.70(0.59)	0.75(0.09)*
South Africa	Aggregate	9.22	6.95**	-52.96(6.94)*	6.42(0.78)*	5.02(10.08)	-0.26(1.14)
	Agriculture	9.08	3.99	27.17(15.12)	-4.76(2.98)	3.50(9.45)	0.20(1.82)
	Manufacturing	9.22	9.88*	46.41(11.08)*	7.31(1.62)*	-0.72(5.15)	0.50(0.74)
	Service	9.22	5.98	-36.61(4.92)*	4.86(0.59)*	3.23(12.02)	-0.06(1.46)

Source: Author

s.e. = standard error; CO2PC = carbon emission per capita; PIMPC = petroleum import per capita. Levels of significance at 1%, 5%, and 10% are represented as *, **, and ***, respectively

With the GDP per capita as the threshold variable, the link between the CO2 emission per capita and the petroleum import per capita is reported in Table 9. In Nigeria, the CO2 emission per capita has positive influence on petroleum import per capita in regime 1 across all sectors, but its effect is either negative or negligible in regime 2. In South Africa, no significant impact of carbon emission per capita is observed in all sectors in both regimes, except in the case of manufacturing where significant effect on petroleum import per capita exists in regime 2. For the reverse effect, petroleum import per capita has significant (positive) effect on carbon emission per capita in Nigeria regime 1 in all sectors except agriculture, whereas this effect is only significant in the manufacturing (negative) and service (positive) sectors. In South Africa, the petroleum import per capita has a significant (positive) impact on carbon emission in regime 1 only in the agricultural sector, but this effect is significant (negative) in all sectors in regime 2, except in agriculture.

These results imply that keeping economic growth in Nigeria below the respective threshold levels reduces petroleum import, following the improvement in environmental quality, which in turn could result from less importation of petroleum products. As the economic growth rises above the threshold, any increase in carbon emission tends to discourage further import of refined petroleum product. In South Africa, the economic growth appears to have no influence on the effect of carbon emission on petroleum import, as this impact largely remains insignificant below or above the threshold levels of growth. However, as the economy grows beyond the threshold, carbon emission tends to fall, which may underscore the role of carbonreducing technology.

Conclusion and policy recommendations

The study investigates the links among refined petroleum import, economic growth, and carbon emission in South Africa and Nigeria both at aggregate and sectoral levels between 1981 and 2015. These links were analyzed using the most recent multivariate Granger causality approach (Bai et al. 2018) and simultaneous equation model (correct for structural breaks in petroleum imports) as well as threshold regression analysis.

The estimates from the simultaneous equation models were used to analyze the possibility of sustainable economic growth in South Africa and Nigeria via the computation of net effects of growth-emission nexus (clean production) and growthenergy import nexus (energy efficiency). There is no evidence of sustainable economic growth in the two countries. In respect of the aggregate economy, although the net effect of growth-emission nexus (which yields clean production) is established for South Africa and Nigeria, considering the structural break, the net effect of growth-petroleum import nexus cannot be estimated for them due to statistically insignificant results. Also, although the net effect of growthpetroleum import nexus (which yields energy inefficiency) is established, without considering the structural break, the net effect of growth-emission nexus cannot be estimated for the countries due to statistically insignificant results.

 Table 9
 Threshold results for the relationship between CO2 emission per capita and petroleum import per capita: threshold variable = GDP per capita

Country	Sector	GDPPC threshold value	Bootstrap	Regime 1 ($\alpha_{it} \leq$	γ)	Regime 2 (α_{it} >	γ)
			LIVI test	Intercept (s.e.)	CO2PC/PIMPC (s.e.)	Intercept (s.e.)	CO2PC/PIMPC (s.e.)
Dependent vari	able, PIMPC; in	dependent variable, CO2F	РС				
Nigeria	Aggregate	7.25	16.50*	-4.07(1.88)*	1.16(0.30)*	8.74(1.48)*	-0.72(0.23)*
	Agriculture	5.57	21.41*	-3.35(1.55)**	1.04(0.24)*	3.87(1.73)*	0.04(0.27)
	Manufacturing	4.80	10.47*	-5.40(1.75)*	1.39(0.27)*	9.05(0.90)*	-0.78(0.14)*
	Service	6.15	18.01*	-3.63(1.54)*	1.09(0.24)*	3.93(1.73)*	0.03(0.27)
South Africa	Aggregate	8.83	18.99*	7.52(8.90)	-0.50(0.97)	2.07(9.98)	0.24(1.09)
	Agriculture	5.20	4.28	-24.53(22.79)	3.09(2.51)	-18.97(47.46)	2.40(5.16)
	Manufacturing	6.81	14.51*	3.57(21.55)	-0.07(2.38)	80.73(18.21)*	8.40(1.99)*
	Service	8.23	20.41*	10.07(8.14)	-0.78(0.88)	10.13(10.81)	-0.65(1.19)
Dependent vari	able, CO2PC; in	dependent variable, PIMF	РС				
Nigeria	Aggregate	7.15	9.86*	4.94(0.41)*	0.36(0.12)*	5.98(0.28)*	0.13(0.07)
	Agriculture	5.44	11.40*	5.80(0.44)*	0.24(0.11)	5.42(0.23)*	0.24(0.06)
	Manufacturing	4.85	9.83*	5.26(0.24)*	0.31(0.06)*	10.13(0.74)*	-0.92(0.19)*
	Service	6.13	14.87*	3.00(0.46)*	0.97(0.12)*	5.63(0.22)*	0.19(0.05)*
South Africa	Aggregate	8.76	11.91*	9.08(0.04)*	0.001(0.02)	9.39(0.03)*	-0.06(0.01)*
	Agriculture	5.20	8.78*	9.03(0.05)*	0.02(0.01)**	9.30(0.08)*	-0.03(0.02)
	Manufacturing	6.81	11.77*	9.08(0.04)*	-0.0004(0.02)	9.37(0.04)*	-0.05(0.01)*
	Service	8.18	6.97**	9.05(0.05)*	0.01(0.01)	9.33(0.07)*	-0.05(0.02)*

Source: Author

s.e. = standard error; CO2PC = carbon emission per capita; PIMPC = petroleum import per capita. Levels of significance at 1%, 5%, and 10% are represented as *, **, and ***, respectively

The results for the agricultural sector show that, for Nigeria, although the net effect of growth-emission nexus indicates a clean production, the net effect of growth-petroleum import nexus suggests energy inefficiency. For South Africa, the net effect of growth-emission nexus does not support a clean production while the net effect of growth-petroleum import nexus cannot be determined as estimates are statistically insignificant.

In the case of manufacturing sector, with or without structural break, the emission-growth nexus for Nigeria generates a positive net effect on output, but the petroleum import-growth nexus yields an increasing net effect on petroleum import. For South Africa, the net effects of emission-growth nexus and petroleum import-growth nexus raise emission and petroleum import, respectively, in the presence of a structural break. Also, an increasing net effect on carbon emission is found (from emission-growth nexus), though an insignificant petroleum import-growth nexus is revealed without a structural break.

Regarding the service sector, the negative net effect of the emission-growth nexus on the environment could not support a clean production while the growth-petroleum import nexus could not be determined in the case of South Africa. Also, estimates could not provide support for both the net effects of emission-growth nexus and growth-petroleum import nexus in respect of Nigeria.

The foregoing findings inform the articulation of the policy implications that would be useful for policy decision-making and managerial insights:

- I. With respect to Nigeria,
- The results suggest that keeping petroleum import per capita above the respective threshold enhances environmental quality as aggregate and sectoral outputs increase. However, the carbon emission can only induce increased output per capita when petroleum import is below the threshold level.
- There is therefore a policy dilemma for managers and policymakers. It implies that managers must strive to achieve an optimal utilization of energy import so as to attain a low carbon development via an increased application of carbon-reducing production techniques as income or output increases. Policymakers should provide profits and import tax rebates to encourage managers to use carbon-reducing production techniques and embark on appropriate pricing of the imported petroleum products.

In the case of South Africa. Π

- Managers and policymakers should note that maintaining petroleum import beyond the threshold may increase the carbon emission per capita associated with a high aggregate output per capita. In turn, such emission may exhibit negligible reverse impact on output per capita.
- Further, setting petroleum import below the threshold value may manage the adverse effect of higher manufacturing output on the environment. However, the results vary across both upper and lower threshold regimes for other sectors.
- Consequently, the managers of the agricultural and service sectors should realize that keeping petroleum import below (or above) the threshold is important as such actions neither raise carbon emission per capita nor reduce output per capita significantly. They should however realize the inefficiency that is associated with increased energy use.

These are inputs into trade (import) and petroleum pricing policy decisions for a sustainable economic growth. Future research efforts should focus on analyzing a sustainable economic growth for other African and developing economies. Further, the implications of other sustainable development goals (such as food security and poverty reduction) for clean production and energy efficiency should be examined.

Acknowledgements We sincerely thank the reviewer and the editor for the insightful and very useful comments and suggestions that have helped to improve the quality of this article. We also thank Mrs. Abosede Adewuyi and Mrs. Adebisi Awodumi for their supports.

Authors' contributions Adeolu. O. Adewuyi: Conceptualization, methodology, software, validation, formal analysis, investigation, resources, data curation, writing-original draft, writing-review and editing, visualization, supervision, project administration, and funding acquisition

Olabanji B. Awodumi: Conceptualization, methodology, software, validation, formal analysis, investigation, resources, data curation, writing-original draft, writing-review and editing, visualization, supervision, project administration, and funding acquisition

Funding No funding support was received from any source for the study.

Data availability The datasets used in the analysis contained in the current study are available in the sources contained in Table 1. They are also available from the corresponding author on reasonable request.

Compliance with ethical standards

Competing interests The authors have no personal or financial competing interests to declare.

Ethics approval and consent to participate None.

Consent for publication None.

N/S	Author	Scope (period)	Estimation techniques	Findings
Oil/Į	setroleum consumption, economic	c growth, and carbon emission nexus		
-	Zou and Chau (2006)	China (1953–2002)	ECM-Granger causality	$OL \rightarrow Y$
0	Halkos and Tzeremes (2011)	42 countries (1986–2006)	DEA and GMM	$OL \rightarrow EFF$
ю	Pereira and Pereira (2010)	Portugal (1977–2003)	VAR (impulse response function and bootstrapping methods)	$OL \rightarrow Y$
4	Wandji (2013)	Cameroon (1971–2009)	ECM	$OL \rightarrow Y$
Ś	Behmiri and Manso (2013)	Sub-Saharan African countries (1985–2011)	FMOLS	$OL \rightarrow Y$ for exporting countries (SR); $OL \leftrightarrow Y$ for importing countries (SR); $OL \leftrightarrow Y$ for both regions (LR)
9	Chu and Chang (2012)	G-6 countries (1971-2010)	Bootstrap panel causality test	$OL \rightarrow Y$ in Germany and Japan; $Y \rightarrow OL$ in the US; neutrality in Canada, France and UK
\sim	Chu (2012)	49 countries (1970–2010)	Bootstrap panel causality test	OL \rightarrow Y for 5 countries; Neutrality for 24 countries; Y \rightarrow OL for 13 countries; and OL \leftrightarrow Y for 7 countries.
8	Behmiri and Manso (2014)	Latin America (1980–2012)	ECM-Granger causality.	

Summary of literature able 10-

Appendix 1

S/N Author	Scope (period)	Estimation techniques	Findings
			Neutral between OL and Y for the Caribbean and South America; $OL \rightarrow Y$ for Central America
9 Akhmat and Zaman (2013)	South Asia (1975–2010)	Bootstrap panel causality test	$OL \rightarrow Y$ for Bhutan, Maldives, Nepal and Sri Lanka; $Y \rightarrow OL$ for Bhutan, Maldives and Sri Lanka
10 Bildirici and Bakirtas (2014)	BRICTS countries (1980-2011)	VECM-Granger causality	In the SR, $Y \rightarrow OL$ in Russia; $OL \rightarrow Y$ in India and Turkey. In the LR, $OL \leftrightarrow Y$ in all countries
11 Lach (2015)	Poland (2000–2009)	Toda-Yamamoto procedure, Bootstrap and VECM-Granger cau-	In the SR, OL \rightarrow Y; In the LR, Y \rightarrow OL
12 Behmiri and Manso (2012a)	27 OECD (1976–2009)	saury VECM-Granger causality	OL↔Y
13 Behmiri and Manso (2012b)	Portugal (1980–2009)	Toda-Yamamoto and VECM-Granger causality	OL↔Y
14 Bildirici and Kayıkçı (2013)	Eurasia (1993–2010)	PMG-Granger causality	OL↔Y
15 Lim et al. (2014)	Philippines (1965–2012)	ECM-Granger causality	OL↔Y
16 Park and Yoo (2014)	Malaysia (1965–2011)	ECM-Granger causality	OL↔Y
17 Yuan et al. (2008)	China (1963–2005)	VECM-Granger causality	OL↔Y
18 Ziramba (2009)	South Africa (1980–2005)	Toda and Yamamoto Granger-causality	OL↔Y
19 Zhang and Yang (2013)	China (1978–2009)	Toda and Yamamoto Granger causality	OL↔Y
20 Alkhathlan and Javid (2013)	Saudi Arabia (1980–2011)	VECM-Granger causality	OL↔Y
21 Saboori and Sulaiman (2013)	Malaysia1 (1980–2009)	VECM-Granger causality	OL↔Y
22 Ali and Harvie (2013)	Libya (1970–2007)	ARDL	$Y \to 0L$
23 Zheng and Luo (2013)	China (1978–2009)	VECM-Granger causality	$\mathbf{Y} \rightarrow \mathbf{OL}$ only in the LR
24 Lean and Smyth (2014)	Malaysia (1980–2011)	ARDL	$Y \rightarrow 0L$
25 Azam (2019)	BRICS (1981–2015)	FMOLS, RLS estimator, and pairwise Dumitrescu-Hurlin panel	EC↔Y
26 Al-Mulali (2011)	MENA countries (1980-2009)	VECM-Granger causality	$OL \leftrightarrow Y$; $OL \leftrightarrow CO2$
27 Shaffiei and Salim (2014)	OECD countries (1980–2011)	ECM-Granger causality	NRE↔C02
28 Dogan and Seker (2016a)	The European Union (1980–2012)	Dynamic panel OLS	$NRE \rightarrow CO2$
29 Dogan and Seker (2016b)	Top renewable energy countries (1985–2011)	FMOLS and dynamic panel OLS	$NRE \rightarrow CO2$
30 Dogan (2015)	Turkey (1990–2012)	VECM-Granger causality	In the short-run, neutrality between NREL and Y; in the long-run, NREL→CO2
31 Dogan (2016)	Turkey (1988–2012)	ARDL	$NRE \rightarrow Y$
32 Bolük and Mert (2014)	16 EU countries (1990-2008)	Pooled OLS	$NRE \rightarrow CO2$
33 Bloch et al. (2012)	China (1965–2008)	VECM	$Y \to C02$
34 Apergis and Payne (2014)	7 Central American countries (1980–2010)	FMOLS and regime-wise Granger causality tests	Y⇔C02
35 Farhani and Shahbaz (2014)	10 MENA countries (1980–2009)	DOLS and FMOLS	$Y \rightarrow CO2 \text{ (short-run)}$
36 Mensah et al. (2019)	22 African countries (1990-2015)	PMG panel ARDL	$CO2 \rightarrow Y$; $FF \leftrightarrow Y$; $FF \leftrightarrow CO2$
37 Dong et al. (2020)	China (1978–2017)	VAR, Granger causality, ANOVA	$Y \rightarrow CO2$

Table 10 (continued)

(continued)
10
ole

Table 10 (continued)			
S/N Author	Scope (period)	Estimation techniques	Findings
38 Adejumo (2019)	Nigeria (1970–2014)	ARDL	$Y \rightarrow CO2$
39 Ali et al. (2020)	33 European Countries (1996–2017)	heterogeneous panel causality and FMOLS	$Y \rightarrow CO2$
40 Balsalobre-Lorente and Leitão	EU-28 (1995–2014)	FMOLS, DOLS, fixed effect and pairwise Dumitrescu-Hurlin	$CO2 \rightarrow Y$
(2020) 41 Rao and Yan (2020)	China (1996–2015)	panel causality LARS-LASSO and stepwise regression	$CO2 \rightarrow Y$
Economic growth as a driver of oil in	aport		
42 Ziramba (2010)	South Africa (1980–2006)	ARDL-ECM	Y has positive effect on CRM
43 Roberts and Rush (2012)	China (1980–2008)	STO	Y has positive effect on NRM
44 Ghosh (2009)	India (1970–2006)	ARDL and Granger causality test	Y has positive effect on CRM
45 Marbuah (2017)	Ghana (1980–2012)	ARDL	Y has positive effect on CRM
46 Kim and Baek (2013)	Korea (1986–2010)	ARDL	Y has positive effect on CRM
47 Camacho-Gutiérrez (2010)	USA (1990–2010)	DOLS	Y has positive effect on CRM
48 Adams et al. (2000)	Thailand (1992–2010)	Simulation	Y has positive effect on CRM
49 Altinay (2007)	Turkey (1980–2005)	ARDL	Y has positive effect on CRM
50 Adewuyi (2016)	Nigeria (1984–2013)	ARDL	Y has positive effect on PTM
51 Zhao and Wu (2007)	China (1995:Q1–2006:Q1)	Johansen and Juselius (1990) co-integration and VECM	Industrial and transport sectors Y have positive effect on OM
52 Mardiana et al. (2013)	Indonesia (1990–2011)	2SLS	Y does not have significant effect on OM
53 Moore (2011)	Barbados (1998–2009)	Pesaran et al. (2001) single equation co-integration approach	Y does not have significant effect on OM
54 Ediger and Berk (2011)	Turkey (1968–2007)	Principal component analysis (PCA) and Euclidean distance	Y does not have significant effect on CRM
55 Balaguer et al. (2015)	Spain (1900–2012)	Incurrotopy Toda-Yamamoto and Johansen Granger causality tests	$EM \to Y$
56 Ozturk and Arisoy (2016)	Turkey (1966–2012)	Time-varying parameters (TVP) approach	Y has positive effect on PTM
V – (real) aross domestic woduct/inc	ome. OI – oil consumution: NBM – nor	ail resonnee imnort. ARDI – antorearessive distributed lea mod	el. ECM – error correction model: OI S – ordinary least

3 Y = (real) gross domestic product/income; OL = oil consumption; NRM = non-oil resource import; ARDL = autoregressive distributed lag model; ECM = square; CRM = crude oil import; DOLS = dynamic OLS; PTM = petroleum import; 2SLS = two-stage least square; OM = oil import; EM = energy import

Appendix 2

Fig. 7 GDP per capita, petroleum import per capita, and carbon emission per capita in Nigeria (Imports of refined petroleum products are measured in liters per capita while GDP and sectoral outputs are the per capita value added constant 2005 US\$).Source: Author's computation; underlying data from USEIA and WDI

Fig. 9 GDP per capita, petroleum import per capita and carbon emission per capita in South Africa (Imports of refined petroleum products are measured in liters per capita while GDP and sectoral outputs are the per capita value added constant 2005 US\$) Source: Author's computation; underlying data from USEIA and WDI

Fig. 8 Sectoral output per capita in Nigeria. Source: Author's computation; underlying data from USEIA and WDI

Fig. 10 Sectoral output per capita in South Africa. Source: Author's computation; underlying data from USEIA and WDI

Appendix 3: Original comprehensive results

Table 11 Conventional unit root test results

	Augmente	d Dickey-Fu	ller (ADF	F) levels	Phillips-Pe	Phillips-Perron (PP)			KPSS			
Levels												
Variables	Constant		Constan	t and trend	Constant		Constant	and trend	Constant		Constant	and trend
	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa
CO2PC	-2.22	-2.46	-2.11	-2.31	-2.22	-2.46	-2.05	-2.31	0.14	0.19	0.13***	0.14
EXR	-2.70***	-2.40	-2.59	-3.40*	-2.24	-2.49	-2.11	-3.52	0.27	0.66**	0.11	0.09
FDEV	-3.06**	-1.55	-3.16	-2.17	-2.32	-1.69	-2.20	-2.18	0.17	0.68**	0.12	0.16
GDPPC	0.44	-0.09	-2.10	-2.04	0.18	-1.31	-2.10	-2.26	0.48**	0.72**	0.20**	0.17
AGDPPC	0.01	-4.00*	-2.11	-1.48	0.01	-4.23*	-2.12	-4.26*	0.64**	0.15	0.15**	0.14
MGDPPC	-0.17	-2.19	-1.02	-2.30	0.11	-2.28	-1.00	-2.26	0.40***	0.17	0.20**	0.16**
SGDPPC	-0.28***	0.80	-1.54	-1.86	1.01	0.43	-1.85	-2.09	0.59**	0.57**	0.19**	0.18**
HCAP	-2.87***	-2.51	-2.73	-2.06	-2.12	-3.67*	-2.15	-1.87	0.23	0.34	0.10	0.17
PCAP	-2.35	-2.88	-3.20	-3.14	-2.66***	-2.37	-3.47	-2.44	0.21	0.18	0.19**	0.17
PIMPC	-2.34	-1.70	-2.79	-3.33***	-2.32	-1.57	-2.78	-2.95	0.33	0.61**	0.14***	0.12
ТО	-1.46	_	-0.32	-	-1.48	-	-1.13	_	0.22	-	0.18**	-
First differe	ence											
CO2PC	-5.623*	-5.70*	-5.27*	-5.70*	-5.87*	-5.70*	-6.63*	-5.73*	0.21	0.12***	0.27*	0.06
EXR	-4.33*	-5.17*	-4.35*	-5.09*	-4.24*	-8.32*	-4.18**	-8.43*	0.10	0.17**	0.05	0.13
FDEV	-5.25*	-6.22*	-5.17*	-6.25*	-8.27*	-6.70*	-8.58*	-10.15*	0.50**	0.24*	0.50*	0.25*
GDPPC	-4.35*	-3.49**	-4.90*	-3.48*	-4.33*	-3.12**	-4.85*	-2.92	0.48**	0.37*	0.09	0.14*
AGDPPC	-5.63*	-6.92*	-5.61*	-6.87*	-5.63*	-10.08*	-5.61*	-8.85*	0.16	0.09	0.08	0.09
MGDPPC	-4.75*	-4.58*	-6.07*	-4.51*	-4.85*	-4.54*	-6.06*	-4.47*	0.50**	0.25	0.07	0.15
SGDPPC	-2.23	-2.83***	-2.18	-3.08	-3.68*	-2.42	-4.50*	-2.14	0.41***	0.43***	0.08	0.14**
HCAP	-5.30*	-5.40*	-5.28*	-5.68*	-5.30*	-5.45*	-5.30*	-7.60*	0.06	0.34*	0.06	0.27*
PCAP	-4.50*	-4.52*	-5.24*	-4.56*	-4.43*	-3.18**	-5.13*	-3.41***	0.46***	0.32*	0.12	0.16**
PIMPC	-8.04*	-5.47*	-7.95*	-5.37*	-8.10*	-6.99*	-7.95*	-6.85*	0.08	0.17**	0.06	0.17**
ТО	-7.33*	_	-7.60*	-	-7.31*	_	-7.57*	_	0.19	_	0.10	_

At levels, the critical values for the ADF test statistics at 1%, 5%, and 10% significance levels are -3.670, -2.964 and -2.621 for the test with constant only, and -4.285, -3.563 and -3.215 for the test with constant and trend, respectively. Critical values are Mckinnon (1996) one-sided *p*-values for the test statistics. The critical values for the PP test statistics at 1%, 5%, and 10% significance levels are -3.662, -2.960 and -2.619 for the test with constant only and -4.285, -3.563, and -3.215 for the test with constant and trend. The critical values for the KPSS test statistics at 1%, 5%, and 10% significance levels are 0.739, 0.463, and 0.347 for the test with constant only while 0.216, 0.146, and 0.119 for the test with constant and trend, respectively

Table 12Conclusion forconventional unit root tests

Variables	ADF		PP	РР		KPSS		Decision	
	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa	Nigeria	South Africa	
CO2PC	I(1)	I(1)	I(1)	I(1)	I(0)	I(1)	I(1)	I(1)	
EXR	I(1)	I(0)	I(1)	I(0)	I(1)	I(1)	I(1)	I(0)	
FDEV	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	
GDPPC	I(1)	I(1)	I(1)	I(1)	I(0)	I(1)	I(1)	I(1)	
AGDPPC	I(1)	I(0)	I(1)	I(0)	I(1)	I(0)	I(1)	I(0)	
MGDPPC	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	
SGDPPC	I(0)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	
HCAP	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)	
PCAP	I(1)	I(1)	I(0)	I(1)	I(0)	I(1)	I(0)	I(1)	
PIMPC	I(1)	I(0)	I(1)	I(1)	I(0)	I(1)	I(1)	I(1)	
ТО	I	_	I(1)	_	1(0)	_	I	_	

The decision made in each of the tests is based on the estimation results with consideration to intercept and trend. I(0) represents stationary of a variable (i.e., significant at level) while I(1) denotes non-stationary (i.e., not significant at level)

Table 13 Nonlinear unit root tests

Variables	<i>KSS^a</i> Nigeria	KSS ^b	KRUSE ^a	KRUSE ^b	<i>KSS^a</i> South Africa	KSS ^b	KRUSE ^a	KRUSE ^b
CO2PC	-1.8484(1)	-0.3442(4)	-1.8619(1)	0.1115(4)	-2.4491(1)	-0.2689(4)	-2.5077(4)	-1.1683(4)
EXR	-1.8573(1)	1.2669(4)	-1.1058(2)	4.5802(4)	-3.7344(1)***	-3.3568(4)**	-4.4519(2)	0.6639(4)
FDEV	-3.0681(1)**	-0.3873(4)	-2.3533(1)	-7.8303(4)	-1.6298(1)	-1.0083(3)	-1.3026(1)	-9.2185(4)
GDPPC	0.1339(1)	1.7027(4)	-0.6421(1)	-6.1919(4)	-0.8103(1)	4.2510(4)***	-0.1610(1)	-2.0749(4)
AGDPPC	0.892(1)	-5.135(-2)***	-0.619(1)	0.853(1)	-2.7210(1)***	-3.269(2)**	-3.707(1)	-0.239(1)
MGDPPC	-2.27(4)*	-5.741(-2)***	-0.950(1)	-4.907(1)	-3.092(1)**	-3.49(4)*	-2.928(1)	-2.36(1)
SGDPPC	2.891(2)*	-4.170(-2)***	-0.402(1)	-4.28(1)	5.16(1)*	4.235(3)*	1.039(1)	2.375(1)
HCAP	-5.0021(4)***	-1.7845(4)	-1.9444(2)	-1.3449(4)	-6.7862(4)***	-1.4760(4)	-1.9589(4)	-2.1929(4)
PCAP	-1.9531(3)	3.0979(3)**	-1.7369(3)	-3.8285(4)	-3.8896(1)***	0.8975(4)	-1.7441(4)	-3.3982(4)
PIMPC	-4.6912(1)***	0.4233(3)	-3.8331(1)	2.9859(3)	-3.0809(1)**	1.9732(4)	-2.4281(2)	-1.6463(3)
ТО	-1.8744(2)	-3.1922(4)**	-2.6837(4)	1.6549(4)	_	_	_	-

KSS and KRUSE with superscript "a" represents de-mean, while with superscript "b" represents de-trended. () is the optimal lag length with the lowest AIC stat. The critical values for the KSS test are -3.48, -2.93, and -2.66 at 1%, 5%, and 10% significance level, respectively. Also, the critical values for the Kruse (2011) test are 13.75, 10.17, and 8.60 at 1%, 5%, and 10% significance level, respectively. *** implies significant at 1%, while ** implies significant at 5% and * significant at 10%

 Table 14
 Conclusion for nonlinear unit root tests

Variables	KSS Nigeri	KRUSE a	Decision	KSS South	KRUSE Africa	Decision
CO2PC	I(1)	I(1)	I(1)	I(1)	I(1)	I(1)
EXR	I(1)	I(1)	I(1)	I(0)	I(1)	I(0)
FDEV	I(0)	I(1)	I(0)	I(1)	I(1)	I(1)
GDPPC	I(1)	I(1)	I(1)	I(0)	I(1)	I(0)
AGDPPC	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
MGDPPC	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
SGDPPC	I(0)	I(0)	I(0)	I(0)	I(0)	I(0)
HCAP	I(0)	I(1)	I(0)	I(0)	I(1)	I(0)
PCAP	I(0)	I(1)	I(0)	I(0)	I(1)	I(0)
PIMPC	I(0)	I(1)	I(0)	I(0)	I(1)	I(0)
ТО	I(0)	I(1)	I(0)	-	-	_

The decision made in each of the tests is based on the estimation results with consideration to both de-mean and de-trended. I(0) represents stationary of a variable (i.e., significant) while I(1) denotes non-stationary (i.e., not significant)

Table 15 Unit root tests with structural breaks (Perron 2006)

Variables	TB Nigeria	t-stat	Decision	TB South Africa	t-stat a	Decision
CO2PC	1998	-4.435816***	I(0)	2001	-4.141879***	I(0)
EXR	1992	-2.299649	I(1)	2009	-3.057859	I(1)
FDEV	2005	-3.998598***	I(0)	1990	-5.029544***	I(0)
GDPPC	2002	-4.929283***	I(0)	1989	-3.194993*	I(0)
AGDPPC	2001	-10.60*	I(0)	1991	-9.837*	I(0)
MGDPPC	1999	-6.29*	I(0)	2008	-5.702*	I(0)
SGDPPC	2009	-5.22*	I(0)	2000	-3.016	I(1)
HCAP	1995	-3.133171*	I(0)	1991	-4.973936***	I(0)
PCAP	2004	-4.918267***	I(0)	2004	-3.555546**	I(0)
PIMPC	1998	-6.499309***	I(0)	2000	-3.663869**	I(0)
ТО	2011	-2.724871	I(1)	_	_	_

The critical values for Perron (2006) are given at the 1%, 5%, and 10% significance levels as -3.9759, -3.4185 and -3.1314. I(0) represents stationary of a variable (i.e., significant at level) while I(1) denotes non-stationary (i.e., not significant at level)

Table 16 Results on the relationship between petroleum import, carbon emission, and aggregate GDP

	Nigeria		South Africa	
Effect of petroleu	m import and carbon emission o	n GDP		
Variable	Model with dummy	Model without dummy	Model with dummy	Model without dummy
CO2PC	-0.2981(0.1126)*	0.2718(0.1199)**	-0.4908(0 .2781)***	-0.3635(0.2678)
PIMPC	-0.0625(0.0325)**	0.1424(0.0359)*	-0.2744(0.0380)	0.1104(0.0145)*
PCAP	-0.0116(0.0260)	-0.2097(0.0695)*	0.5623(0.1225)*	0.1564(0.0691)**
HCAP	0.0586(0.1533)	-0.02587(0.2028)	-0.0037(0.0474)	-0.0143(0.0588)
FDEV	_	_	_	-0.4576(0.2594)***
PIMPDU	0.1287(0.0421)*	_	0.1616(0 .0458)*	-
С	0.0371(0.6512)	1.7893(1.0551)***	7.1923(0.4459)*	8.0342(0.3675)*
Hansen	7.242	16.175	10.9790	7.3753
Effect of petroleu	m import and GDP on carbon er	nission		
GDPPC	-2.9302(1.0012)*	0.6306(0.9605)	-2.0051(0.9187)**	-1.6138(1.1307) ***
PIMPC	-0.1801(0.0716)*	-0.1616(0.0918)***	-0.0589 (0.0573)	0.1668(0.1122)
PCAP	-0.0673(0.1217)	0.1316(1.1888)	1.1399(0.4739)**	0.4017(0.3031)
ТО	-0.1799(2879)	-0.4892(0.2653)***	_	-
FDEV	_	_	0.0082(0.0645)	-0.8478(0.44672)***
PIMPDU	0.3575(0.1196)*	_	0.3281(0.1343)**	-
С	0.8390(0.6282)	-0.1498(0.8090)	14.3627(6.7398)**	12.4655(8.7701)
Hansen	7.242	16.175	10.9790	7.3753
Effect of carbon e	emission and GDP on petroleum	import		
GDPPC	-10.9815(6.6881)	5.4428(2.4867)**	-1.5727(8.2040)	7.3024(1.3533)*
CO2PC	-4.1154(2.1444)***	-1.6746(0.9015)***	-3.2567(5.2507)	1.8204(3.4419)*
EXR	0.2218(0.8061)	0.1140(0.4835)	10.1718(14.2220)	-0.7791(08211)
М	0.3363(0.7074)	0.4818(0.5583)	-4.8777(7.6479)	4.4388(3.3394)
PCAP	_	1.1417(0.5413)**	_	-
PIMPDU	1.5100(0.6987)**	_	4.1434 (4.8319)	_
С	2.0098(1.7009)	-3.8352(2.1739)***	-31.2896(28.4698)	-57.2886(13.4459)*
Hansen	7.242	16.175	10.9790	7.3753

Source: Author's computation; *,**, and*** denote significance at 1%, 5%, and 10% levels, respectively; values in parentheses are the standard errors

Table 17	Results on the relationshi	p between	petroleum import	t, carbon emission,	, and agricultural GDP

	Nigeria		South Africa	
Effect of petroleum	import and carbon emission on agricu	ltural GDP		
Variable	Model with dummy	Model without dummy	Model with dummy	Model without dummy
CO2PC	-0.6728(0.2026)*	2.8822(0.7286)*	0.9402(0.0.8377)	-1.0393(0.2696)*
PIMPC	0.0396(0.1051)	0.7708(0.2023) *	0.1794(0.1038)***	-0.0969(0.0391*
PCAP	-0.0280(0.1442)	-0.0223(0.1510)	-0.2854(0.3956)	0.5931(0.2217)*
HCAP	-0.3729(0.5147)	-0.2373(0.5415)	-0.7619(0.3466)**	0.0131(0.1544)
TO	-0.8174(0.3970)**	0.0479(0.4219)		_
FDEV	_	_	-0.5626(0.5962)	-1.2232(0.3663)*
PIMPDU	0.8138(0.1364)*	_	-0.2015(0.1086)***	_
С	7.0093(2.2200)*	4.2217(2.4596)**	9.0103(2.3592)*	3.7341(1.0711)*
Hansen	11.792	12.820	6.2879	12.484
Effect of petroleum	import and agricultural GDP on carbo	on emission		
AGDPPC	-1.0932(0.4274)*	0.3377(0.0737)*	-0.0045(0.2262)	-0.95198(0.4525)**
PIMPC	-0.0086(0.1302)	-0.2613(0.0790)*	-0.0405(0.035)**	-0.0899(0.0508)***
PCAP	0517(0.1939)	0.0043(0.0532)	0.0096(0.1439)	0.5676(0.3166)***
TO	-0.9854(0.6052)***	0.0029(0.1470)	_	
FDEV	_	_	-0.2928(0.3333)	-1.1548(0.5789) **
PIMPDU	0.9741(0.3781)*	_	0.0626(0.0496)	_
С	6.0668(2.4562)*	-1.0428(0.4352)	-0.1078(0.9676)	3.5954(1.7780)**
Hansen	11.792	12.820	6.2879	12.484
Effect of carbon em	ission and agricultural GDP on petrole	eum import		
AGDPPC	1.4380(1.0810)	1.3041(0.2986)*	-1.5788(5.0667)	1.9745(2.4590)
CO2PC	-0.2375(0.8790)	-3.8460(1.1690)*	-14.4493(7.7507)***	-1.0575(3.2864)
EXR	-0.33536(0.7501)	0.0069(0.3067)	7.2751(8.4803)	-4.9232(1.3287)*
FDEV	1.1436(0.4576)*	-0.0095(0.2942)	-8.1376(10.6143)	4.6879(4.3291)
PIMPDU	-0.6454(0.9533)	_	3.2199(2.0914)	_
С	-7.3974(5.7730)	-3.9465(2.0593)**	-23.2058 (25.8516)	15.7398(10.3023)
Hansen	11.792	12.820	6.2879	12.484

Source: Author's computation; *, **, and*** denote significance at 1%, 5%, and 10% levels, respectively; values in parentheses are the standard errors

Table 18	Results on the relationship	between	petroleum impo	ort, carbon e	emission, an	d manufacturing	GDP
				,			

	Nigeria		South Africa	
Effect of petroleum	import and carbon emission on manuf	acturing GDP		
Variable	Model with dummy	Model without dummy	Model with dummy	Model without dummy
CO2PC	2.2379(0.5893)*	2.1554(0.5229)*	0.6643(0.1864)*	-0.6320(0.3752)***
PIMPC	0.5723(0.1634)*	0.5550(0.1311)*	0.1029(0.0402)*	0.0051(0.0208)
PCAP	0.1174(0.1109)	0.1168(0.0987)	0.1299(0.1162)	0.3143(0.1147)*
HCAP	-0.1774(0.3121)	-0.1638(0.2822)	-0.0201(0.0971)	-0.0429(0.1179)
ТО	-0.0662(0.1948)	-0.0575(0.1788)	_	_
FDEV	_	_	-0.8157(0.2717)*	-0.7069(0.3009)**
PIMPDU	-0.0312(0.2486)	_	-0.1400(0.0624) **	_
С	2.9227(1.4519)**	2.9090(1.3628)**	6.2878(0.5845)*	6.1199(0.7009)*
Hansen	10.292	11.376	13.553	9.2641
Effect of petroleum	import and manufacturing GDP on car	rbon emission		
MGDPPC	0.4167(0.1559)*	0.4362(0.1484)*	1.2386(0.3664)*	-1.0897(0.3541)*
PIMPC	-0.2583(0.0764)*	-0.2605(0.0786)*	-0.1581(0.0344)*	-0.0021(0.0240)
PCAP	-0.0278(0.0566)	-0.0302(0.0532)	-0.0845(0.1456)	0.3467(0.2002)***
ТО	0.0085(0.0832)	0.0049(0.0782)	_	_
FDEV	_	_	1.1119(0.4426)*	-0.9616(0.2995)*
PIMPDU	0.0152(0.1322)	_	0.2170(0.0603)*	6.4700(2.0441)*
С	-0.9266(0.6494)	-0.9908(0.6033)***	-7.8128(2.324)*	_
Hansen	10.292	11.376	13.553	9.2641
Effect of carbon emi	ission and manufacturing GDP on petr	oleum import		
MGDPPC	1.4778(0.7287)**	1.5642(0.6691)**	6.2442(2.7840)**	6.0765(2.7902)**
CO2PC	-3.8090(1.7265)**	-3.7408(1.990)**	-6.5205(1.7332)*	-2.7509(5.8099)
EXR	0.0727(0.3015)	0.1043(0.3010)	0.9697(3.2737)	-4.4311(0.9960)*
FDEV	-0.0721(0.2463)	-0.0822(0.2478)	5.8316(3.5583)***	4.3803(4.8081)
PIMPDU	0.0878(0.5885)	_	1.6321(0.9703)***	_
С	-3.3372(3.4972)	-3.6736(3.3484)	-44.6464(15.4808)*	-17.8791(19.5350)
Hansen	10.292	11.376	13.553	9.2641

Source: Author's computation; *, **, and*** denote significance at 1%, 5%, and 10% levels, respectively; values in parentheses are the standard errors

Table 19 Results on the relationship between petroleum import, carbon emission, and service GDP

	Nigeria		South Africa	
Effect of petroleu	im import and carbon emission or	n service GDP		
Variable	Model with dummy	Model without dummy	Model with dummy	Model without dummy
CO2PC	0.7520(0.3582)**	0.4799(0.6422)	-0.3003(0.0558)*	0.3695(0.0751)*
PIMPC	0.2252(0.1676)	0.4521(0.1082)*	0.0158(0.0140)	0.0187(0.0054)*
PCAP	0.6250(0.2201)*	0.2484(0.1980)	-0.0779(0.0442)***	-0.0623(0.0373)***
HCAP	-1.9196(1.0132)**	-3.5817(0.6836)*	0.0013(0.0152)	0.0109(0.0310)
ТО	0.5677(0.6736)	-0.6494(0.5290)	_	-
FDEV	-	_	-0.1425(0.0815)***	-0.0109(0.0365)
PIMPDU	0.4367(0.2804)	_	0.0010(0.0186)	_
С	10.7372(5.1337)**	19.6447(3.3690)*	0.1777(0.1406)	-0.0782(0.2132)
Hansen	11.761	11.588	19.776	19.037
Effect of petroleu	im import and service GDP on ca	rbon emission		
SGDPPC	0.5081(0.6961)	0.0781(0.1280)	2.8508(0.4907)*	2.3403(0.5814)*
PIMPC	-0.0973(0.1255)	-0.1658(0.0819)**	-0.0842(0.0409)**	-0.0419(0.0169)*
PCAP	-0.5216(0.6520)	-0.0783(0.1117)	0.2553(0.1299)**	0.1852(0.0893)**
ТО	-1.10430.9394)	-0.3864(0.2503)	_	-
FDEV	-0.0146(0.3503)	0.0402(0.1942)	0.5076(0.2565)**	0.0294(0.1007)
PIMPDU	-0.3876(0.6872)	_	0.0495(0.0573)	-
С	0.0823(0.9238)	0.3868(0.8314)	-0.5127(0.3063)***	-0.4157(0.2338)***
Hansen	11.761	11.588	19.776	19.037
Effect of carbon	emission and service GDP on pet	roleum import		
SGDPPC	-0.0181(0.5769)	0.3744(0.4030)	-4.7031(8.7579)	5.8049(17.3444)
CO2PC	-3.6741(1.3957)*	-3.3923(1.3035)*	-2.8784(2.9935)	1.1153(6.5515)
EXR	1.2726(0.6777)***	0.8652(0.6316)	1.6119(1.6123)	-4.0686(2.0185)**
FDEV	0.4271(0.5447)	0.8344(0.5450)	0.1735(2.7083)	2.4608(1.9776)
PIMPDU	0.5310(0.5674)	_	1.7977(0.4492)*	_
С	2.2968(3.8590)	-0.0200(2.6734)	-4.7919(7.5234)	22.0184(9.3600)**
Hansen	11.761	11.588	19.776	19.037

Source: Author's computation; *, **, and *** denote significance at 1%, 5%, and 10% levels, respectively; values in parentheses are the standard errors

References

- Ackah I (2015) On the relationship between energy consumption, productivity and economic growth: evidence from Algeria, Ghana, Nigeria and South Africa. SSRN Electron J. https://doi.org/10. 2139/ssrn.2614493
- Adams FG, Ichino Y, Prazmowski PA (2000) Economic growth and energy import requirements: an energy balance model of Thailand. J Policy Model 22(2):219–254
- Adejumo OO (2019) Environmental quality vs economic growth in a developing economy: complements or conflicts. Environ Sci Pollut Res 27:6163–6179. https://doi.org/10.1007/s11356-019-07101-x
- Adewuyi AO (2016) Determinants of import demand for non-renewable energy (petroleum) products: empirical evidence from Nigeria. Energy Policy 95:73–93
- Adewuyi AO, Awodumi OB (2017a) Renewable and non-renewable energy-growth-emissions linkages: review of emerging trends with policy implications. Renew Sust Energ Rev 69:275–291
- Adewuyi AO, Awodumi OB (2017b) Biomass energy consumption, economic growth and carbon emissions: fresh evidence from West Africa using a simultaneous equation model. Energy 119:453–471

- Adewuyi AO, Awodumi OB (2020) Environmental pollution, energy resource import, economic growth and financial development: theoretical exploration and empirical evidence from Nigeria. Int J Environ Sci Nat Resour 26(1):21–31
- Akhmat G, Zaman K (2013) Nuclear energy consumption, commercial energy consumption and economic growth in South Asia: bootstrap panel causality test. Renew Sust Energ Rev 25:552–559
- Aklin M (2016) Re-exploring the trade and environment nexus through the diffusion of pollution. Environ Resour Econ 64:663–682
- Ali I, Harvie C (2013) Oil and economic development: Libya in the post-Gaddafi era. Econ Model 32:273–285
- Ali M, Raza SA, Khamis B (2020) Environmental degradation, economic growth, and energy innovation: evidence from European countries. Environ Sci Pollut Res 27:28306–28315. https://doi.org/10.1007/ s11356-020-09142-z
- Alkhathlan K, Javid M (2013) Energy consumption, carbon emissions and economic growth in Saudi Arabia: an aggregate and disaggregate analysis. Energy Policy 62:1525–1532
- Aller C, Ductor L, Herrerias MJ (2015) The world trade network and the environment. Energy Econ 52:55–68
- Al-Mulali U (2011) Oil consumption, CO2 emission and economic growth in MENA countries. Energy 36:6165–6171

- Altinay G (2007) Short-run and long-run elasticities of import demand for crude oil in Turkey. Energy Policy 35:5829–5835
- Apergis N, Payne JE (2014) Renewable energy, output, CO2 emissions, and fossil fuel prices in Central America: evidence from a nonlinear panel smooth transition vector error correction model. Energy Econ 42:226–232
- Apergis N, Jebli MB, Youssef SB (2018) Does renewable energy consumption and health expenditures decrease carbon dioxide emissions? Evidence for sub-Saharan Africa countries. Renew Energy 127:1011–1016
- Awodumi OB, Adewuyi AO (2020) The role of non-renewable energy consumption in economic growth and carbon emission: evidence from oil producing economies in Africa. Energy Strategy Rev 27: 1–19
- Azam M (2019) Relationship between energy, investment, human capital, environment, and economic growth in four BRICS countries. Environ Sci Pollut Res 26:34388–34400. https://doi.org/10.1007/ s11356-019-06533-9
- Baek EG, Brock WA (1992) A nonparametric test for independence of a multivariate time series. Stat Sin 2(1):137–156
- Bah MM, Abdulwaki MM, Azam M (2020) Income heterogeneity and the environmental Kuznets curve hypothesis in Sub-Saharan African countries. GeoJournal 85:617–628
- Bahmani-Oskooee M, Economidou C (2009) Export led growth vs. growth led exports: LDCs experience. J Dev Areas 42(2):179–212
- Bai ZD, Wong WK, Zhang BZ (2010) Multivariate linear and non-linear causality tests. Math Comput Simul 81:5–17
- Bai ZD, Li H, Wong WK, Zhang BZ (2011) Multivariate causality tests with simulation and application. Stat Probabil Lett 81(8):1063–1071
- Bai ZD, Hui Y, Jiang D, Lv Z, Wong W, Zheng S (2018) A new test of multivariate nonlinear causality. PLoS One 13(1):e0185155. https:// doi.org/10.1371/journal.pone.0185155
- Balaguer J, Florica T, Ripolless J (2015) Foreign trade and economic growth in Spain (1900–2012): the role of energy imports. Econ Polit 32:359–375
- Balsalobre-Lorente D, Leitão NC (2020) The role of tourism, trade, renewable energy use and carbon dioxide emissions on economic growth: evidence of tourism-led growth hypothesis in EU-28. Environ Sci Pollut Res. https://doi.org/10.1007/s11356-020-10375-1
- Barro R. J. (2003). Determinants of economic growth in a panel of countries. Annals of Economics and Finance, 4:231–274
- Behmiri NB, Manso JRP (2012a) Crude oil conservation policy hypothesis in OECD (organisation for economic cooperation and development) countries: a multivariate panel Granger causality test. Energy 43:253–260
- Behmiri NB, Manso JRP (2012b) Does Portuguese economy support crude oil conservation hypothesis? Energy Policy 45:628–634
- Behmiri NB, Manso JRP (2013) How crude oil consumption impacts on economic growth of Sub-Saharan Africa? Energy 54:74–83
- Behmiri NB, Manso JRP (2014) The linkage between crude oil consumption and economic growth in Latin America: the panel framework investigations for multiple regions. Energy 72:233–241
- Bildirici ME, Bakirtas T (2014) The relationship among oil, natural gas and coal consumption and economic growth in BRICTS (Brazil, Russian, India, China, Turkey and South Africa) countries. Energy 65:134–144
- Bildirici ME, Kayıkçı F (2013) Effects of oil production on economic growth in Eurasian countries: panel ARDL approach. Energy 49: 156–161
- Bloch H, Rafiq S, Salim R (2012) Coal consumption, CO2 emission and economic growth in China: empirical evidence and policy responses. Energy Econ 34:518–528
- Bolük G, Mert M (2014) Fossil & renewable energy consumption, GHGs (greenhouse gases) and economic growth: evidence from a panel of EU (European Union) countries. Energy 74(1):439–446

- Boubellouta B, Kusch-Brandt S (2020) Testing the environmental Kuznets curve hypothesis for E-waste in the EU28+2 countries. J Clean Prod 277:123371. https://doi.org/10.1016/j.jclepro.2020. 123371
- Bretschger L (2005) Economics of technological change and the natural environment: how effective are innovations as a remedy for resource scarcity? Ecol Econ 54:148–163
- Busse M, Groizard JL (2008) Technology trade in economic. World Econ 31(4):569–592
- Camacho-Gutiérrez P (2010) Dynamic OLS estimation of the US import demand for Mexican crude oil. J Int Bus Econ 11, 2, 75–82; MPRA Paper No. 30608
- Cherniwchan J (2017) Trade liberalization and the environment: evidence from NAFTA and US manufacturing. J Int Econ 105:130–149
- Chu H (2012) Oil consumption and output: what causes what? Bootstrap panel causality for 49 countries. Energy Policy 51:907–915
- Chu H, Chang T (2012) Nuclear energy consumption, oil consumption and economic growth in G-6 countries: bootstrap panel causality test. Energy Policy 48:762–769
- Churchill SA, Inekwe J, Ivanovski K, Smyth R (2020) The environmental Kuznets curve across Australian states and territories. Energy Econ 90:104869. https://doi.org/10.1016/j.eneco.2020.104869
- Cleveland CJ, Costanza RK, Hall CAS, Kaufmann RK (1984) Energy and the US economy: a biophysical perspective. Science 225:890– 897
- Copeland BR, Taylor MS (2005) Free trade and global warming: a trade theory view of the Kyoto protocol. J Environ Econ Manag 49:205– 234
- Dasgupta S, Roy J (2015) Understanding technological progress and input price as drivers of energy demand in manufacturing industries in India. Energy Policy 83:1–13
- Dogan E (2015) The relationship between economic growth and electricity consumption from renewable and non-renewable sources: a study of Turkey. Renew Sust Energ Rev 52:534–546
- Dogan E (2016) Analyzing the linkage between renewable and nonrenewable energy consumption and economic growth by considering structural break in time-series data. Renew Energy 99:1126– 1136
- Dogan E, Seker F (2016a) Determinants of CO2 emissions in the European Union: the role of renewable and non-renewable energy. Renew Energy 94:429–439
- Dogan E, Seker F (2016b) The influence of real output, renewable and non-renewable energy, trade and financial development on carbon emissions in the top renewable energy countries. Renew Sust Energ Rev 60:1074–1085
- Dong B, Xu Y, Fan X (2020) How to achieve a win-win situation between economic growth and carbon emission reduction: empirical evidence from the perspective of industrial structure upgrading. Environ Sci Pollut Res. https://doi.org/10.1007/s11356-020-09883-x
- Ediger VS, Berk I (2011) Crude oil import policy of Turkey: historical analysis of determinants and implications since 1968. Energy Policy 39:2132–2142
- Fang G, Lu L, Tian L, He Y, Bai Y (2020) Can China achieve the energysaving and emission reducing objectives during the "13th five-yearplan"?—a systematic evolutionary analysis. J Clean Prod 262:1–12
- Farhani S, Shahbaz M (2014) What role of renewable and non-renewable electricity consumption and output is needed to initially mitigate CO2 emissions in MENA region? Renew Sust Energ Rev 40:80–90
- Fedoseeva S, Zeidan R (2018) How (a)symmetric is the response of import demand to changes in its determinants? Evidence from European energy imports. Energy Econ 69:379–394
- Ghosh S (2009) Import demand of crude oil and economic growth: evidence from India. Energy Policy 37:699–702

- Goldberg P, Khandelwal A, Pavcnik N, Topalova P (2010) Imported intermediate inputs and domestic product growth: evidence from India. Q J Econ 125(4):1727–1767
- Goldstein M, Khan MS (1985) Chapter 20 income and price effects in foreign trade. Handb Int Econ 2:1041–1105
- Grossman GM, Krueger A (1995) Economic growth and the environment. Q J Econ 110(2):353–377
- Halkos GE, Tzeremes NG (2011) Oil consumption and economic efficiency: a comparative analysis of advanced, developing and emerging economies. Ecol Econ 70:1354–1362
- Hall C, Tharakan P, Hallock J, Cleveland C, Jefferson M (2003) Hydrocarbons and the evolution of human culture. Nature 426: 318–322
- Halliru AM, Loganathan N, Hassan AAG, Mardani A, Kamyab H (2020) Re-examining the environmental Kuznets curve hypothesis in the economic community of West African states: a panel quantile regression approach. J Clean Prod 276:124247. https://doi.org/10. 1016/j.jclepro.2020.124247
- Halpern L, Koren M, Szeidl A (2015) Imported inputs and productivity. Am Econ Rev 105(12):3660–3703
- Hansen BE (1999) Threshold effects in non-dynamic panels: estimation, testing, and inference. J Econ 93(2):345–368
- Hansen BE (2000) Sample splitting and threshold estimation Author(s). Econometrica 68(3):575–603
- Hayashi F (2000) Econometrics. Princeton University Press, Princeton
- Hiemstra C, Jones JD (1994) Testing for linear and nonlinear Granger causality in the stock price-volume relation. J Financ 49(5):1639– 1664
- Hu ZG, Yuan JH, Hu Z (2011) Study on China's low carbon development in an economy-energy-electricity-environment framework. Energy Policy 39(5):2596–2605
- Huo T, Li X, Cai W, Zuo J, Jia F, Wei H, (2020) The impact of urbanization on urban building carbon emissions in China: an empirical study based on provincial panel data. Sustain Cities Soc (2020). 10.1016/j.scs.2020.102068, 1-40
- IPCC, 2018: Summary for Policymakers. In: Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty [Masson-Delmotte, V., P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield (eds.)]. In Press.
- Johansen S, Juselius K (1990) Maximum likelihood estimation and inference on cointegration — with applications to the demand for money. Oxf Bull Econ Stat 52(2):169–210
- Jones LE, Manuelli LE (1997) The sources of growth. J Econ Dyn Control 21(1):75–114
- Kasahara H, Rodrigue J (2008) Does the use of imported intermediates increase productivity? Plant-level evidence. J Dev Econ 87(1):106– 118
- Kim HS, Baek J (2013) Assessing dynamics of crude oil import demand in Korea. Econ Model 35:260–263
- Kruse R (2011) A new unit root test against ESTAR based on a class of modified statistics. Stat Pap 52:71–85
- Lach L (2015) Oil usage, gas consumption and economic growth: evidence from Poland. Energy Sources, Part B: Economics, Planning, and Policy 10(3):223–232
- Lean HH, Smyth R (2014) Disaggregated energy demand by fuel type and economic growth in Malaysia. Appl Energy 132:168–177
- Lee J (1995) Capital goods imports and long-run growth. J Dev Econ 48(1):91–110

- Lim K, Lim S, Yoo S (2014) Oil consumption, CO2 emission, and economic growth: evidence from the Philippines. Sustainability 6:967– 979
- Lin B, Agyeman SD (2019) Assessing Ghana's carbon dioxide emissions through energy consumption structure towards a sustainable development path. J Clean Prod 238:117941. https://doi.org/10.1016/j. jclepro.2019.117941
- Ma X, Wang C, Dong B, Gu G, Chen R, Li Y, Zou H, Zhang W, Li Q (2019) Carbon emissions from energy consumption in China: its measurement and driving factors. Sci Total Environ 648:1411–1420
- Maji IK (2015) Does clean energy contribute to economic growth? Evidence from Nigeria. Energy Rep 1:145–150. https://doi.org/10. 1016/j.egyr.2015.06.001
- Marbuah G (2017) Understanding crude oil import demand behaviour in Africa: the Ghana case. Journal of African Trade 4(1–2):75–87
- Mardiana DA, Husin Z, Hamzah MZ, Kartoatmodjo RT (2013) Economy growth and oil import requirement in Indonesia. J. Energy Resour. Technol. Policy 3(11):76–83
- Mensah JT, Marbuah G, Amoah A (2016) Energy demand in Ghana: a disaggregated analysis. Renew Sust Energ Rev 53:924–935
- Mensah IA, Sun M, Gao C, Omari-Sasu AY, Zhu D, Ampimah BC, Quarcoo A (2019) Analysis on the nexus of economic growth, fossil fuel energy consumption, CO2 emissions and oil price in Africa based on a PMG panel ARDL approach. J Clean Prod 228:161–174
- Moore A (2011) Demand elasticity of oil in Barbados. Energy Policy 39: 3515–3519
- Narayan P, Smyth R (2005) Electricity consumption, employment and real income in Australia evidence from multivariate Granger causality tests. Energy Policy 33(9):1109–1116
- Olper A, Curzi D, Raimondi V (2017) Imported intermediate inputs and firms' productivity growth: evidence from the food industry. J Agric Econ 68(1):280–300
- Omer AM (2008) Energy, environment and sustainable development. Renew Sust Energ Rev 12(9):2265–2300
- Omri A (2014) An international literature survey on energy-economic growth nexus: evidence from country-specific studies. Renew Sust Energ Rev 38:951–959
- Oyedepo SO (2012) Energy and sustainable development in Nigeria: the way forward. Energy Sustain Soc 2(15):1–17
- Ozturk I (2010) A literature survey on energy–growth nexus. Energy Policy 38:340–349
- Ozturk I, Arisoy I (2016) An estimation of crude oil import demand in Turkey: evidence from time-varying parameters approach. Energy Policy 99:174–179
- Pao H, Chen C (2018) Decoupling strategies: CO emissions, energy resources, and economic growth in the Group of Twenty. J Clean Prod 2018:907–919. https://doi.org/10.1016/j.jclepro.2018.09.190
- Park S, Yoo S (2014) The dynamics of oil consumption and economic growth in Malaysia. Energy Policy 66:218–223
- Payne JE (2009) On the dynamics of energy consumption and output in the US. Appl Energy 86:575–577
- Pereira AM, Pereira RMM (2010) Is fuel-switching a no-regrets environmental policy? VAR evidence on carbon dioxide emissions, energy consumption and economic performance in Portugal. Energy Econ 32:227–242
- Perron P (2006) Dealing with structural breaks. In: Patterson K, Mills TC (eds) Palgrave handbook of econometrics, Vol. 1. Palgrave Macmillan, London, pp 278–352
- Pesaran MH, Shin Y, Smith RJ (2001) Bounds testing approaches to the analysis of level relationships. J Appl Econ 16:289–326
- Porter ME, van der Linde C (1995) Toward a new conception of the environment competitiveness relationship. J Econ Perspect 9(4): 97–118
- Porter ME, van der Linde C (1999) Green and competitive: ending the stalemate. Journal of Business Administration and Politics:215–230

- Rao C, Yan B (2020) Study on the interactive influence between economic growth and environmental pollution. Environ Sci Pollut Res 27:39442–39465. https://doi.org/10.1007/s11356-020-10017-6
- Roberts I, Rush A (2012) Understanding China's demand for resource imports. China Econ Rev 23(3):566–579
- Romer D (1996) Advanced macroeconomics. McGraw-Hill
- Rosen MA (2009) Energy sustainability: a pragmatic approach and illustrations. Sustainability 2009(1):55–80. https://doi.org/10.3390/ su1010055
- Saboori M, Sulaiman J (2013) Environmental degradation, economic growth and energy consumption: evidence of the environmental Kuznets curve in Malaysia. Energy Policy 60:892–905
- Safarzadeh S, Rasti-Barzoki M, Hejazi SR (2020) A review of optimal energy policy instruments on industrial energy efficiency programs, rebound effects, and government policies. Energy Policy 139: 111342. https://doi.org/10.1016/j.enpol.2020.111342
- Shafiei S. and Salim R. A. (2014). Non-renewable and renewable energy consumption and CO2 emissions in OECD countries: A comparative analysis. Energy Policy, 66:547–556
- Shahbaz M, Lean HH, Farooq A (2013) Natural gas consumption and economic growth in Pakistan. Renew Sust Energ Rev 18:87–94
- Solow RM (1993) An almost practical step toward sustainability. Resour Policy 19:162–172
- Solow RM (1997) Reply: Georgescu-Roegen versus Solow/Stiglitz. Ecol Econ 22:267–268
- Stern DI (1997) Limits to substitution and irreversibility in production and consumption: a neoclassical interpretation of ecological economics. Ecol Econ 21:197–215
- Tajudeen IA (2015) Examining the role of energy efficiency and noneconomic factors in energy demand and CO2 emissions in Nigeria: policy implications. Energy Policy 86:338–350
- United Nations (1987). Our common future. Report of the World Commission on Environment and Development, United Nations
- US Energy Information Administration (USEIA) (2010). International Energy Outlook (2010) US Department of Energy. Washington, DC
- Wandji YDF (2013) Energy consumption and economic growth: evidence from Cameroon. Energy Policy 61:1295–1304
- World Summit on Sustainable Development (2002). Environment and sustainable development: implementation of Agenda 21 and the

Programme for the Further Implementation of Agenda 21. United Nations General Assembly, Draft resolution submitted by the Vice-Chairman of the Committee, Mr. Jan Kára (Czech Republic), on the basis of informal consultations held on draft resolution A/C.2/57/L.83

- Yang HL, Tian LX, Ding ZW (2004) Sustainable economic growth under restraint of energy. Syst Eng 22(3):40–43
- Yao S, Zang S, Zang X (2019) Renewable energy, carbon emission and economic growth: a revised environmental Kuznets curve perspective. J Clean Prod 235:1338–1352
- Yu B, Li YL, Chi CJ (2006) Endogenous model with exhaustible energy, pollution abatement spending and sustainable growth induced. Journal of Management Science in China 9(4):12–17
- Yuan J, Kang J, Zhao C, Hu Z (2008) Energy consumption and economic growth: evidence from China at both aggregated and disaggregated levels. Energy Econ 30:3077–3094
- Zhang W, Yang S (2013) The influence of energy consumption of China on its real GDP from aggregated and disaggregated viewpoints. Energy Policy 57:76–81
- Zhao X, Wu Y (2007) Determinants of China's energy imports: an empirical analysis. Energy Policy 35(8):4235–4246
- Zheng Y, Luo D (2013) Industrial structure and oil consumption growth path of China: empirical evidence. Energy 57:336–343
- Ziramba E (2009) Disaggregate energy consumption and industrial production in South Africa. Energy Policy 37:2214–2220
- Ziramba E (2010) Price and income elasticities of crude oil import demand in South Africa: a cointegration analysis. Energy Policy 38: 7844–7849
- Zou G, Chau KW (2006) Short- and long-run effects between oil consumption and economic growth in China. Energy Policy 34:3644– 3655
- Zuo H, Ai D (2011) Environment, energy and sustainable economic growth. Proc Eng 21:513–519

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.